

Vol. 11 (2023) (/2227-9091/11) Vol. 9 (2021) (/2227-9091/10) Vol. 9 (2021) (/2227-9091/9) Vol. 7 (2019) (/2227-9091/8) Vol. 7 (2019) (/2227-9091/7) Vol. 6 (2018) (/2227-9091/6) Vol. 5 (2017) (/2227-9091/6) Vol. 4 (2016) (/2227-9091/4) Vol. 3 (2015) (/2227-9091/4) Vol. 2 (2014) (/2227-9091/2) Vol. 1 (2013) (/2227-9091/1)

Risks, Volume 11, Issue 1 (January 2023) – 21 articles



Cover Story (view full-size image (/files/uploaded/covers/risks/big_cover-risks-v11-i1.png)): Integrated thinking is a strategic response by business leaders, encompassing aspects such as setting up a sustainability committee and environmental and social reporting, pursuing the United Nations Sustainable Development Goals, and obtaining assurance for integrated reports. The authors sought to answer the question: Can integrated thinking decrease financial risk? The answer is positive. In a global sample of 7111 companies over five years, integrated thinking was deemed to increase company liquidity (proxied by the cash ratio) and decrease the weighted average cost of capital. These relationships are further nuanced by considering the compensation of the chief executive officer. View this paper (https://www.mdpi.com/2227-9091/11/1/6)

(https://www.mdpi.com/2227-

905974971969 regarded as officially published after their release is announced to the table of contents alert mailing list (/journal/risks/toc-alert).

- You may sign up for e-mail alerts (/journal/risks/toc-alert) to receive table of contents of newly released issues.
- PDF is the official format for papers published in both, html and pdf forms. To view the papers in pdf format, click on the "PDF Full-text" link, and use the free <u>Adobe Reader (http://www.adobe.com/)</u> to open them.

Order results	
Publication Date	
Result details	
Compact	
Show export options ~	
Open Access Article	
Valuation of Equity-Linked Death Benefits on Two Lives with Dependence (/2227-9091/11/1/21)	
Risks 2023, 11(1), 21; https://doi.org/10.3390/risks11010021 (https://doi.org/10.3390/risks1101002	<u>21)</u> - 12 Jan 2023
Open Access Article	
Adversarial Artificial Intelligence in Insurance: From an Example to Some Potential Remedies (<u>/2227-9091/11/1/20)</u>
Risks 2023, 11(1), 20; https://doi.org/10.3390/risks11010020 (https://doi.org/10.3390/risks1101002	<u>20)</u> - 11 Jan 2023
Open Access Article	<u> </u>
FAANG Stocks, Gold, and Islamic Equity: Implications for Portfolio Management during COVID-	<u>-19 (/2227-9091/11/1/19)</u>
Risks 2023, 11(1), 19; https://doi.org/10.3390/risks11010019 (https://doi.org/10.3390/risks1101001	<u>19)</u> - 11 Jan 2023
Open Access Feature Paper Article	
Dependence Modelling of Lifetimes in Egyptian Families (/2227-9091/11/1/18)	
Risks 2023, 11(1), 18; https://doi.org/10.3390/risks11010018 (https://doi.org/10.3390/risks1101001	<u>18)</u> - 11 Jan 2023
Open Access Article	
Rational versus Irrational Behavior of Indonesian Cryptocurrency Owners in Making Investmen	<u>nt Decision (/2227-9091/11/1/17)</u>
Risks 2023, 11(1), 17; https://doi.org/10.3390/risks11010017 (https://doi.org/10.3390/risks1101001	<u>17)</u> - 11 Jan 2023
Open Access Article	
Trade Credit Management and Profitability of Jordanian Manufacturing Firms (/2227-9091/11/1/1	<u>16)</u>
Risks 2023, 11(1), 16; https://doi.org/10.3390/risks11010016 (https://doi.org/10.3390/risks1101001	<u>16)</u> - 10 Jan 2023
Open Access Article	
A Wavelet Analysis of the Dynamic Connectedness among Oil Prices, Green Bonds, and CO2 E	<u>missions (/2227-9091/11/1/15)</u>
Risks 2023 11(1) 15: https://doi.org/10.3390/risks11010015 (https://doi.org/10.3390/risks110100	15) - 09 Jan 2023

Opén-Access Article	₽	<u> </u>
Analysis of Yields and Their Determinants in the European Corporate Green Bond Market (/2	27-90	<u>091/11/1/14)</u>
<i>Rist</i> 023, <i>11</i> (1), 14; <u>https://doi.org/10.3390/risks11010014 (https://doi.org/10.3390/risks11010</u>	<u>014)</u> -	- 06 Jan 2023 <u>k a (noggle_desktop_layout_cookie)</u> Q =
Open Access Article	≡	(/2227-9091/11/1/13/pdf?version=1673000617)
Risk Measures in Simulation-Based Business Valuation: Classification of Risk Measures in R Practice (/2227-9091/11/1/13)	isk A	xiom Systems and Application in Valuation
Risks 2023, 11(1), 13; https://doi.org/10.3390/risks11010013 (https://doi.org/10.3390/risks11010	<u>013)</u> -	- 06 Jan 2023
Open Access Article	≡	
Regulating Robo-Advisors in Insurance Distribution: Lessons from the Insurance Distribution	n Dire	<u>ctive and the AI Act (/2227-9091/11/1/12)</u>
Risks 2023, 11(1), 12; https://doi.org/10.3390/risks11010012 (https://doi.org/10.3390/risks11010	<u>012)</u> -	- 04 Jan 2023
Open Access Review	₽	<u> </u>
Customer Due Diligence in the FinTech Era: A Bibliometric Analysis (/2227-9091/11/1/11)		
<i>Risks</i> 2023, <i>11</i> (1), 11; <u>https://doi.org/10.3390/risks11010011 (https://doi.org/10.3390/risks11010</u>	<u>011)</u> -	03 Jan 2023
Open Access Editorial	₽	<u> </u>
<u>Continuing Risks (/2227-9091/11/1/10)</u>		
<i>Risks</i> 2023, <i>11</i> (1), 10; <u>https://doi.org/10.3390/risks11010010 (https://doi.org/10.3390/risks11010</u>	<u>010)</u> -	- 27 Dec 2022
Open Access Article	≡	
Economic Value Added Research: Mapping Thematic Structure and Research Trends (/2227-	091/1	<u>1/1/9)</u>
Risks 2023, 11(1), 9; https://doi.org/10.3390/risks11010009 (https://doi.org/10.3390/risks110100	<u>09)</u> - 2	26 Dec 2022
Open Access Article	≡	
Opportunities for the Application of a Model of Cost Management and Reduction of Risks in OLAP Technology: The Case of the Agro-Industrial Sector of Russia (/2227-9091/11/1/8)	inano	cial and Economic Activity Based on the
Risks 2023, 11(1), 8; https://doi.org/10.3390/risks11010008 (https://doi.org/10.3390/risks110100	<u>08)</u> - 2	23 Dec 2022
Open Access Article	≣	
<u>Deep Generators on Commodity Markets Application to Deep Hedging (/2227-9091/11/1/7)</u>		
Risks 2023, 11(1), 7; https://doi.org/10.3390/risks11010007 (https://doi.org/10.3390/risks110100	<u>07)</u> - 2	23 Dec 2022
Open Access Article	≡	
The Relationship between Integrated Thinking and Financial Risk: Panel Estimation in a Glob	al Sai	<u>mple (/2227-9091/11/1/6)</u>
Risks 2023, 11(1), 6; https://doi.org/10.3390/risks11010006 (https://doi.org/10.3390/risks110100	<u>06)</u> - 2	23 Dec 2022
Open Access Article	≡	
Relationship between Complex Integration Indices and Inflation Indicators and Their Impact of	on the	Development of Regional Cooperation
Detween Countries to Reduce the Level of Inhationary Risks: Case of the SCO member Countries 2023, 11(1), 5: https://doi.org/10.3300/risks110100	<u>ries (</u> 05) - ($\frac{12227 - 3031/11/1/5)}{22 \text{ Dec } 2022}$
1000 2020, 11(1), 0, <u>mtps.//doi.org/10.0050/13k511010000 (mtps.//doi.org/10.0050/13k5110100</u>	<u>00)</u> - /	-
Open Access Feature Paper Article	₽	
ECLIPSE: Holistic AI System for Preparing Insurer Policy Data (/2227-9091/11/1/4)		
<i>Risks</i> 2023, 11(1), 4; <u>https://doi.org/10.3390/risks11010004 (https://doi.org/10.3390/risks110100</u>	<u>04)</u> - 2	21 Dec 2022
Open Access Article	≡	
Methodology for Economic Analysis of Highly Uncertain Innovative Projects of Improbability	<u>Type</u>	<u>(/2227-9091/11/1/3)</u>
Risks 2023, 11(1), 3; https://doi.org/10.3390/risks11010003 (https://doi.org/10.3390/risks110100	<u>03)</u> - 2	20 Dec 2022
Open Access Article	≡	
The Role of Emotions and Knowledge on Preference for Uncertainty: Follow Your Heart but L	isten	<u>to Your Brain! (/2227-9091/11/1/2)</u>
Risks 2023, 11(1), 2; https://doi.org/10.3390/risks11010002 (https://doi.org/10.3390/risks110100	<u>02)</u> - 2	20 Dec 2022
Open Access Article	≡	
Recursive Approaches for Multi-Layer Dividend Strategies in a Phase-Type Renewal Risk Mo	<u>del (/2</u>	227-9091/11/1/1 <u>)</u>
Risks 2023, 11(1), 1; https://doi.org/10.3390/risks11010001 (https://doi.org/10.3390/risks110100	<u>01)</u> - 2	20 Dec 2022
Show export options ~		

Previous Issue Volume 10, December (/2227-9091/10/12)

H

<u>Risks (/journal/risks)</u>, EISSN 2227-9091, Published by MDPI <u>RSS (/rss/journal/risks)</u> <u>Content Alert (/journal/risks/toc-alert)</u>

Further Information

Article Processing Charges (/apc) Pay an Invoice (/about/payment) Open Access Policy (/openaccess) Contact MDPI (/about/contact) Jobs at MDPI (https://careers.mdpi.com)

Guidelines

 For Authors (/authors)

 For Reviewers (/reviewers)

 For Editors (/editors)

 For Librarians (/librarians)

 For Publishers (/publishing_services)

 For Societies (/societies)

 For Conference Organizers (/conference_organizers)

 MDPI Initiatives

 Sciforum (https://sciforum.net)

 MDPI Books (https://www.mdpi.com/books)

 Preprints.org (https://www.preprints.org)

 Scilit (https://www.scilit.net)

 SciProfiles (https://sciprofiles.com)

 Encyclopedia (https://encyclopedia.pub)

 JAMS (https://jams.pub)

 Proceedings Series (/about/proceedings)

Follow MDPI

LinkedIn (https://www.linkedin.com/company/mdpi) Facebook (https://www.facebook.com/MDPIOpenAccessPublishing) Twitter (https://twitter.com/MDPIOpenAccess)

Subscribe to receive issue release notifications and newsletters from MDPI journals

Select options

Enter your email address...

Subscribe

© 1996-2023 MDPI (Basel, Switzerland) unless otherwise stated

Disclaimer Terms and Conditions (/about/terms-and-conditions) Privacy Policy (/about/privacy)



!	<u> Journals (/about/journals)</u>	<u>Topics (/topics)</u>	Information (/authors)	Author Services (/authors/english)	(<u>/toggldnitistives (labeut/icitistive</u> sQ =
					<u>About (/about)</u>
	Sign In / Sign	Jp (/user/login)	Submit (https:/	/susy.mdpi.com/user/manuscr	ipts/upload?journal=risks)
Search for	r Articles:				
Title / Key	word				
Author / Af	filiation / Email				
Risks					
All Article T	ypes				
			Search		
<u>Advanced</u>	Search				
Journals (/al	<u>bout/journals)</u> / <u>Risks (/</u>	ournal/risks) / Edit	orial Board	Tracked for Impact	2.2
` 🥒 ''	ISNS	<u>_(journal/noico)</u>		Factor	2.2

Submit to Risks (https://susy.mdpi.com/user/manuscripts/upload?form[journal_id]=162)

Review for Risks (https://susy.mdpi.com/volunteer/journals/review)

Journal Menu

Journal Menu

- <u>Risks Home (/journal/risks)</u>
- Aims & Scope (/journal/risks/about)
- Editorial Board (/journal/risks/editors)
- <u>Reviewer Board (/journal/risks/submission_reviewers</u>)
- <u>Topical Advisory Panel (/journal/risks/topical_advisory_panel)</u>
- Instructions for Authors (/journal/risks/instructions)
- <u>Special Issues (/journal/risks/special_issues)</u>
- Topics (/topics?query=&journal=risks&status=all&category=all)
- <u>Topical Collections (/journal/risks/topical_collections)</u>
- <u>Article Processing Charge (/journal/risks/apc)</u>
- Indexing & Archiving (/journal/risks/indexing)
- Most Cited & Viewed (/journal/risks/most_cited)
- Journal Statistics (/journal/risks/stats)
- Journal History (/journal/risks/history)
- Journal Awards (/journal/risks/awards)
- <u>Editorial Office (/journal/risks/editorial_office)</u>
- <u>10th Anniversary of Risks (/journal/risks/anniversary)</u>

Journal Browser

Journal Browser

volume

issue

> Forthcoming issue (/2227-9091/11/3)

> Current issue (/2227-9091/11/2)

Vol. 11 (2023) (/2227-9091/11) Vol. 9 (2021) (/2227-9091/10) Vol. 9 (2021) (/2227-9091/9) Vol. 7 (2019) (/2227-9091/8) Vol. 7 (2019) (/2227-9091/7) Vol. 6 (2018) (/2227-9091/6) Vol. 5 (2017) (/2227-9091/6) Vol. 4 (2016) (/2227-9091/4) Vol. 3 (2015) (/2227-9091/4) Vol. 2 (2014) (/2227-9091/2) Vol. 1 (2013) (/2227-9091/1)

[™] <u>(/toggle_desktop_layout_cookie)</u> Q ≡



Editorial Board

Members (78)

Search by first name, last name, affiliation,



Prof. Dr. Mogens Steffensen

<u>Website (https://sites.google.com/site/steffensenmogens/)</u> <u>SciProfiles (https://sciprofiles.com/profile/93295)</u> Editor-in-Chief

Department of Mathematical Sciences, University of Copenhagen, Universitetsparken 5, DK-2100 Copenhagen Ø, Denmark Interests: life insurance mathematics; asset-liability management; optimal asset allocation; personal finance and insurance; stochastic control theory Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Corina Constantinescu

Website (https://www.liverpool.ac.uk/mathematical-sciences/staff/dana-constantinescu/)

SciProfiles (https://sciprofiles.com/profile/175608)

Associate Editor

Institute for Financial and Actuarial Mathematics, Department of Mathematical Sciences, University of Liverpool, Liverpool L69 7ZL, UK Interests: actuarial science; risk theory; dependence structures; heavy-tailed distributions; bonus-malus systems Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Montserrat Guillén

Wessite (http://www.ub.edu/riskcenter/guillen/) SciProfiles (https://sciprofiles.com/profile/294352)

Associate Editor

Department of Econometrics, Riskcenter-IREA Universitat de Barcelona Av. Diagonal, 690 08034 Barcelona, det extreme desktop layout cookie, $Q \equiv$ **Interests:** risk; insurance; actuarial statistics; long-term care insurance; experience rating; statistical methods for insurance and finance, automobile fraud detection, quantitative methods for risk management; longevity; pension-saving investment; risk analytics

Special Issues, Collections and Topics in MDPI journals

Dr. S. Nuray Akin

Website (http://www.nurayakin.com/)

Editorial Board Member

Marshall School of Business, University of Southern California, FBE, Bridge Hall 308, 3670 Trousdale Pkwy, Los Angeles, CA 90089, USA Interests: macroeconomics; social insurance; risk management

Dr. Hamed Amini

Website (http://amini.gsucreate.org/)

Editorial Board Member

Department of Risk Management and Insurance, J. Mack Robinson College of Business, Georgia State University, Atlanta, GA 30303, USA **Interests:** random structures; systemic risk; credit risk; liquidity risk; financial networks



Prof. Dr. Séverine Arnold

Website (http://hec.unil.ch/people/sarnold) SciProfiles (https://sciprofiles.com/profile/1217404)

Editorial Board Member

Department of Actuarial Science, Faculty of Business and Economics (HEC Lausanne), University of Lausanne, 1015 Lausanne, Vaud, Switzerland **Interests:** mortality models; cause-of-death mortality rates; cause-specific mortality dependence; social security; notional defined contribution pension schemes

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Benjamin Avanzi

<u>Website (https://www.business.unsw.edu.au/our-people/benjaminavanzi)</u> <u>SciProfiles (https://sciprofiles.com/profile/196288)</u> Editorial Board Member

School of Risk and Actuarial Studies, UNSW Australia Business School, UNSW Sydney, Sydney, NSW 2052, Australia Interests: insurance capital modelling (reserving, solvency, dependence); risk theory (optimal control of actuarial surplus models); social security and pensions; risk modelling in operations management

Prof. Dr. Florin Avram

Website (http://avram.perso.univ-pau.fr/) SciProfiles (https://sciprofiles.com/profile/377435)

Editorial Board Member

Laboratoire de Mathématiques Appliquées, Université de Pau, 64013 Pau CEDEX, France

Interests: stochastic processes; risk; mathematical finance; inventory; queueing and population dynamics

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Mercedes Ayuso <u>Website (http://www.ub.edu/riskcenter/ayuso/)</u> <u>SciProfiles (https://sciprofiles.com/profile/156523)</u> Editorial Board Member

Riskcenter, Universidad de Barcelona, Av. Diagonal, 690, 08034 Barcelona, Spain

Interests: pensions; longevity; ageing and insurance; actuarial science; risks and insurance; long-term care insurance; pricing, statistical methods for automobile fraud detection; quantitative methods for BI; traffic crashes; telematics

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Gurdip Bakshi <u>Website (http://scholar.rhsmith.umd.edu/gbakshi/Home)</u> Editorial Board Member Department of Finance, Robert H. Smith School of Business, 4413 Van Munching Hall, University of Maryland, College Park, MD 20742-1815, USA Interest stock valuation; option valuation; term structure of interest rates; asset pricing; capital and currency markets; crashes; default risk; density approximations; aging; heterogeneity in beliefs; volatility; international finance

Prof. Dr. Alejandro Balbás

Website (http://www.uc3m.es/portal/page/portal/dpto_economia_empresa/home/faculty/alejandrobalbas) SciProfiles (https://sciprofiles.com/profile/92892)

Editorial Board Member

Department of Business Administration, University Carlos III of Madrid, C/ Madrid, 126, 28903 Getafe, Madrid, Spain

Interests: risk management; asset pricing; fixed income

Special Issues, Collections and Topics in MDPI journals

Prof. Dr. Francesca Barigozzi

Website (https://sites.google.com/site/francescabarigozziunibo/home)

Editorial Board Member

Department of Economics, University of Bologna, P.zza Scaravilli 2, 40126 Bologna, Italy

Interests: health insurance; insurance contracts (moral hazard and adverse selection); behavioral decision theory; information economics; health economics; industrial organization



Dr. Daniel Bauer

Website (http://robinson.gsu.edu/profile/daniel-bauer/)

Editorial Board Member

Department of Risk Management and Insurance, J. Mack Robinson College of Business, Georgia State University, 35 Broad Street, 11th Floor, Atlanta, GA 30303, USA

Interests: actuarial science; life insurance mathematics; financial mathematics; risk management

Prof. Dr. Brendan K Beare

Website (https://www.sydney.edu.au/arts/about/our-people/academic-staff/brendan-beare.html)

SciProfiles (https://sciprofiles.com/profile/963163)

Editorial Board Member

School of Economics, University of Sydney, Sydney, NSW 2006, Australia Interests: pricing kernel; payoff distribution pricing model; copula

Prof. Dr. Fred Espen Benth

Website (https://www.mn.uio.no/math/personer/vit/fredb/) SciProfiles (https://sciprofiles.com/profile/836580)

Editorial Board Member

Department of Mathematics, University of Oslo, 0316 Blindern, Norway

Interests: stochastic analysis; mathematical finance; energy and commodity markets

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Enrico Biffis

Website (http://www.imperial.ac.uk/people/e.biffis) SciProfiles (https://sciprofiles.com/profile/74184)

Editorial Board Member

Finance Department, Imperial College Business School, Imperial College London, South Kensington Campus, London SW7 2AZ, UK Interests: risk management; insurance; asset-liability management; alternative risk transfers; catastrophe risk; actuarial science

Prof. Dr. M. Martin Boyer

Website (http://www.hec.ca/profs/martin.boyer.html) SciProfiles (https://sciprofiles.com/profile/40148)

Editorial Board Member

Department of Finance, HEC Montréal, Université de Montréal 3000, Chemin de la Côte-Sainte-Catherine, Montréal, QC H3T 2A7, Canada Interests: insurance economics; corporate risk management; contract theory; financial risk management; property and casualty insurance markets



Prof. Dr. Jorge Miguel Bravo

<u>Website (https://www.novaims.unl.pt/magic/index.php/component/k2/item/313-jorge-bravo-biografia.html)</u> SciProfiles (https://sciprofiles.com/profile/1124588)

Editorial Board Member

- 1. NOVA IMS—Information Management School, New University of Lisbon, 1070-312 Lisbon, Portugal
- 2. Department of Economics, Université Paris-Dauphine PSL, 75775 Paris, France

Interests: longevity risk management; pensions; actuarial science; social policy; financial economics; labour economics; interest rate risk management; creckersk management; data science; population economics

Special Issues, Collections and Topics in MDPI journals



∑<u>(/toggle_desktop_layout_cookie)</u> Q <u>≡</u>

Prof. Dr. Andrew J. G. Cairns

Website (http://www.macs.hw.ac.uk/~andrewc/) SciProfiles (https://sciprofiles.com/profile/2231597)

Editorial Board Member

Actuarial Mathematics and Statistics, School of Mathematical and Computer Sciences, Heriot-Watt University, Edinburgh EH14 4AS, UK **Interests:** actuarial science; insurance mathematics; longevity risk; mortality risk; quantitative risk management

Prof. Dr. Roy Cerqueti

Website (https://web.uniroma1.it/disse/node/7444) SciProfiles (https://sciprofiles.com/profile/1348786)

Editorial Board Member

Department of Social and Economic Sciences, Sapienza University of Rome, 00185 Rome, Italy

Interests: rank-size analysis; econophysics; statistical and quantitative modelling for finance; long-term memory; applied probability; complex networks; copula theory

Special Issues, Collections and Topics in MDPI journals

Prof. Dr. Carolyn Chang

Website (http://business.fullerton.edu/Department/Finance/?ID=ChanCarol#FinanceFacultyInfo?user=cchang)

Editorial Board Member

Department of Finance, Mihaylo College of Business and Economics, Califonia State University Fullerton, Fullerton, CA 92834, USA Interests: catastrophe risk management and modeling; insurance and weather derivatives; forecasting VIX and pricing of VIX products; Asia-Pacific financial market



Prof. Dr. Arthur Charpentier

Website (https://freakonometrics.github.io/) SciProfiles (https://sciprofiles.com/profile/166693)

Editorial Board Member

Département de mathématiques, Université du Québec à Montréal, Montreal, QC H2X 3Y7, Canada

Interests: actuarial science; predictive modeling; computational science; statistics and econometrics; risks; visualization; data science Special Issues, Collections and Topics in MDPI journals

Dr. Hua Chen

<u>Website (http://www.fox.temple.edu/mcm_people/dr-hua-chen/)</u> <u>SciProfiles (https://sciprofiles.com/profile/2231661)</u> Editorial Board Member

Department of Risk, Insurance and Healthcare Management, Fox School of Business, Temple University, Philadelphia, PA 19122, USA **Interests:** insurance economics; corporate risk management; systemic risk; catastrophic risk modeling and pricing; alternative risk transfer



Dr. Eric Cheung

Website (https://www.unsw.edu.au/staff/eric-c-k--cheung) SciProfiles (https://sciprofiles.com/profile/193276)

Editorial Board Member

School of Risk and Actuarial Studies, UNSW Business School, UNSW, Sydney, NSW 2052, Australia

Interests: insurance risk theory; ruin theory; aggregate claims analysis; credibility; financial mathematics; matrix-analytic methods; queueing theory; operations research; applied probability

Special Issues, Collections and Topics in MDPI journals

Dr. Albert Cohen

SciProfiles (https://sciprofiles.com/profile/135859)

Editorial Board Member

Department of Mathematics, Michigan State University, East Lansing, MI 48824, USA

Interests: financial mathematics; financial markets; actuarial science; insurance; financial risk management

Special Issues, Collections and Topics in MDPI journals



Editorial Board Member

Dipartine di Scienze Economiche, Aziendali e Statistiche. Viale delle Scienze, Edificio 13, 90128 Palermo, Italy Interests: financial and insurance modeling; portfolio optimization; numerical methods for risk management; computational finance; simulation models Spectra Issues, Collections and Topics in MDPI journals

Prof. Dr. J. David Cummins

Website (http://fox.temple.edu/mcm_people/dr-j-david-cummins)

Editorial Board Member

Department of Risk, Insurance, and Healthcare Management, Fox School of Business, Temple University, Philadelphia, PA 19122, USA **Interests:** insurance economics; efficiency; data envelopment analysis; systemic risk; securitization; organizational form; economies of scope

Dr. Michel Dacorogna

Website (https://www.bf.uzh.ch/en/persons/dacorogna-michel/teaching) SciProfiles (https://sciprofiles.com/profile/590189)

Editorial Board Member

Prime Re Solutions, 6340 Zug, Switzerland

Interests: quantitative risk management; finance; insurance economics; actuarial science

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Angelos Dassios

Website (http://stats.lse.ac.uk/angelos/) SciProfiles (https://sciprofiles.com/profile/88932)

Editorial Board Member

Department of Statistics, London School of Economics, Houghton Street, London WC2A 2AE, UK

Interests: insurance mathematics; ruin theory; path dependent options; point processes; financial mathematics; excursion theory

Special Issues, Collections and Topics in MDPI journals

Prof. Dr. Georges Dionne

Website (http://chairegestiondesrisques.hec.ca/en/) SciProfiles (https://sciprofiles.com/profile/125925)

Editorial Board Member

Department of Finance, HEC Montréal, 3000, Chemin de la Côte-Sainte-Catherine, Montréal, QC, Canada

Interests: risk management for private and social risks; microeconomic theory under uncertainty (financial contracts, insurance contracts); asymmetric information (moral hazard and adverse selection); economics of health services; regulation in transportation and the environment

Special Issues, Collections and Topics in MDPI journals



Dr. Ugo Fiore

Website (http://wpage.unina.it/ufiore/) SciProfiles (https://sciprofiles.com/profile/128367)

Editorial Board Member

Department of Management and Quantitative Sciences, Parthenope University of Naples, Naples, Italy

Interests: data science; optimization; security; deep learning; energy-aware systems

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. José Garrido

<u>Website (http://www.concordia.ca/artsci/math-stats/faculty.html?fpid=jose-garrido)</u> <u>SciProfiles (https://sciprofiles.com/profile/63040)</u> Editorial Board Member

Department of Mathematics and Statistics, Concordia University, 1455 de Maisonneuve Blvd West, LB-921.21, Montreal, QC H3G 1M8, Canada Interests: risk theory; insurance statistics; credibility theory; risk measures; actuarial and financial mathematics

Special Issues, Collections and Topics in MDPI journals

Prof. Dr. Nadine Gatzert

Website (https://www.vwrm.rw.fau.de/team/prof-dr-nadine-gatzert/) SciProfiles (https://sciprofiles.com/profile/107563)

Editorial Board Member

Institute of Insurance Economics and Risk Management, Friedrich-Alexander-University Erlangen-Nürnberg, Lange Gasse 20, D-90403 Nürnberg, Germany

Interests: life insurance mathematics; alternative risk transfer; valuation and management of financial guarantees; enterprise risk management; modeling and management of mortality and longevity risk; regulation and solvency assessment

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Paolo Giudici

<u>K X (/toggle_desktop_layout_cookie)</u> Q ≡

Website (https://sites.google.com/a/unipv.it/giudici/) SciProfiles (https://sciprofiles.com/profile/479650)

Editorial Board Member

Department of Economics and Management, University of Pavia, 27100 Pavia, Italy

Interests: financial data science; graphical models; network models; financial networks; systemic risk; financial risk management; fintech risk management

Special Issues, Collections and Topics in MDPI journals

Prof. Dr. Emilio Gómez Déniz

Website (https://dmc.ulpgc.es/emilio-gomez-deniz.html) SciProfiles (https://sciprofiles.com/profile/337163)

Editorial Board Member

Department of Quantitative Methods and TIDES Institute, University of Las Palmas de Gran Canaria, Campus de Tafira s/n, 35017 Las Palmas, Spain **Interests:** distributions theory; bayesian statistics; robustness; bayesian applications in economics (actuarial, credibility, ruin theory) <u>Special Issues, Collections and Topics in MDPI journals</u>



Prof. Dr. Steven Haberman

Website (https://www.bayes.city.ac.uk/faculties-and-research/experts/steven-haberman)

SciProfiles (https://sciprofiles.com/profile/1153224)

Editorial Board Member

Faculty of Actuarial Science and Insurance, Bayes Business School, City, University of London, 106 Bunhill Row, London EC1Y 8TZ, UK Interests: actuarial science; applied statistics; insurance; modelling of mortality trends and longevity risk; pensions and annuity modelling Special Issues, Collections and Topics in MDPI journals

Dr. Marius Hofert

Website (https://saasresearch.hku.hk/~mhofert/) SciProfiles (https://sciprofiles.com/profile/247853)

Editorial Board Member

Department of Statistics and Actuarial Science, The University of Hong Kong, Hong Kong, China

Interests: dependence modeling; computational statistics; data science; quantitative risk management

Special Issues, Collections and Topics in MDPI journals

Prof. Dr. Dayong Huang

Website (https://sites.google.com/a/uncg.edu/dayong-huang/) SciProfiles (https://sciprofiles.com/profile/93064)

Editorial Board Member

Department of Accounting and Finance, Bryan School of Business and Economics, University of North Carolina Greensboro, Greensboro, NC 27412, USA

Interests: investments; macroeconomics; asset pricing



Prof. Dr. Sebastian Jaimungal

Website (http://www.utstat.utoronto.ca/sjaimung)

Editorial Board Member

Department of Statistical Sciences, University of Toronto, 100 St. George Street, Toronto, ON M5S 3G3, Canada Interests: applied stochastic control; algorithmic and high frequency trading; ambiguity aversion; financial engineering; financial insurance



Prof. Dr. Krzysztof Jajuga

Website (https://www.ue.wroc.pl/pracownicy/krzysztof_jajuga.html) SciProfiles (https://sciprofiles.com/profile/1316277)

Editorial Board Member

Department of Financial Investments and Risk Management, Wroclaw University of Economics and Business, ul. Komandorska 118/120, 53-345 Wroclaw, Poland

Interests: multivariate data analysis; classification; econometrics; financial markets; risk management; machine learning <u>Special Issues, Collections and Topics in MDPI journals</u>

Dr. Lanpeng Ji

<u>Website (https://eps.leeds.ac.uk/maths/staff/5350/dr-lanpeng-ji)</u> Editorial Board Member



Dr. Meelis Käärik

<u>Website (https://www.etis.ee/Portal/Persons/Display/c1add49a-9da1-4691-a075-a6a0596f82e5?tabld=CV_ENG)</u> <u>SciProfiles (https://sciprofiles.com/profile/203509)</u>

Editorial Board Member

Institute of Mathematics and Statistics, University of Tartu, Tartu, Estonia

Interests: premium estimation and reserving in non-life insurance; approximation of distributions; skewed distributions

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Dimitris Karlis

<u>Website (http://www2.stat-athens.aueb.gr/~karlis/)</u> <u>SciProfiles (https://sciprofiles.com/profile/949206)</u> Editorial Board Member

Department of Statistics, Athens University of Economics and Business, 10434 Athens, Greece

Interests: mixture models; EM algorithm; distribution theory; sports statistics and modelling; Copulas; multivariate count data; discrete valued time series



Prof. Dr. Rüdiger Kiesel

<u>Website (http://www.lef.wiwi.uni-due.de/team/ruediger-kiesel/)</u> <u>SciProfiles (https://sciprofiles.com/profile/174372)</u> Editorial Board Member

Chair for Energy Trading and Finance, University Duisburg-Essen, Universitätsstraße 12, 45141 Essen, Germany

Interests: energy derivatives; quantitative climate finance; financial risk management; financial derivatives

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Tomas Kliestik

<u>Website (https://scholar.google.com/citations?user=zsFNpOgAAAAJ&hl=en)</u> <u>SciProfiles (https://sciprofiles.com/profile/745651)</u> Editorial Board Member

Faculty of Operation and Economics of Transport and Communications, University of Zilina, Univerzitna 1, 010 26 Zilina, Slovakia

Interests: financial management; behavioral finance; econometric models; credit risk

Special Issues, Collections and Topics in MDPI journals

Prof. Dr. Stéphane Loisel

Website (http://isfaserveur.univ-lyon1.fr/~stephane.loisel/) SciProfiles (https://sciprofiles.com/profile/61152)

Editorial Board Member

ISFA, Université Lyon 1, 50 avenue Tony Garnier, F-69007 Lyon, France

Interests: risk management; insurance; ruin theory; Solvency II; economic capital; entreprise risk management; longevity risk; customer behaviour in insurance

Prof. Dr. Brenda López-Cabrera

<u>Website (http://www.wiwi.hu-berlin.de/professuren/quantitativ/statistik/members/personalpages/bl/Publications/home)</u> Editorial Board Member

C.A.S.E. Centre for Applied Statistics and Economics, School of Economics and Business Administration, Humbolt Universität zu Berlin, Unter den Linden 6, 10099 Berlin, Germany

Interests: electricity, energy, weather, agricultural markets; quantitative climate finance; insurance and finance; financial derivatives; financial risk management; empirical and computational finance; dimension reduction techniques; extreme value modeling



Prof. Dr. Elisa Luciano
<u>Website (http://sites.carloalberto.org/luciano/)</u> <u>SciProfiles (https://sciprofiles.com/profile/2500612)</u> Editorial Board Member Department of Economics and Statistics, University of Torino, Corso Unione Sovietica 218 bis, I-10134 Torino, Italy Interest: Josk management; credit risk; dependence in financial markets; markets with frictions; insurance applications



<u>{X} (/toggle_desktop_layout_cookie)</u> Q ≡

Dr. Xavier Milhaud

Website (http://www.xaviermilhaud.fr/en/index-en.html) SciProfiles

Editorial Board Member

ISFA, Lab of Actuarial and Financial Sciences, University Lyon 1, Lyon, France

Interests: actuarial science; risk aggregation; stress tests in insurance; parametric and nonparametric statistical models; policyholder's behaviours

Prof. Dr. Jens Perch Nielsen

<u>Website (https://www.cass.city.ac.uk/faculties-and-research/experts/jens-perch)</u> <u>SciProfiles (https://sciprofiles.com/profile/488156)</u> Editorial Board Member

Cass Business School, City, University of London, 106 Bunhill Row, London EC1Y 8TZ, UK Interests: machine learning in insurance; structured nonparametric statistics; pension research

Special Issues, Collections and Topics in MDPI journals

Prof. Dr. Annamaria Olivieri

<u>Website (https://mib.edu/en/school/faculty/annamaria-olivieri-faculty-mib)</u> <u>SciProfiles (https://sciprofiles.com/profile/205887)</u> Editorial Board Member

Department is Economics and Management, University of Parma, Via Kennedy 6, 43100 Parma, Italy

Interests: risk management for life insurance and pension funds, in particular with reference to longevity risk; solvency for life portfolios and pension funds; actuarial perspectives of annuitization and post-retirement choices in pension products; multistate models for the insurances of the person; actuarial pricing of life and health insurance products; actuarial models for the valuation of the life insurance business

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Cornelis W. Oosterlee

Website (https://www.cwi.nl/people/kees-oosterlee) SciProfiles (https://sciprofiles.com/profile/518603)

Editorial Board Member

Delft Institute of Applied Mathematics, TU Delft, 2628 XE Delft, The Netherlands and CWI-Centrum Wiskunde & Informatica, 1098 XE Amsterdam, The Netherlands

Interests: risk management; computational finance; scientific computing; applied mathematics; numerical mathematics

Special Issues, Collections and Topics in MDPI journals

Prof. Dr. Zbigniew Palmowski

Website (http://prac.im.pwr.wroc.pl/~zpalma/index.html) SciProfiles (https://sciprofiles.com/profile/584736)

Editorial Board Member

Department of Applied Mathematics, Faculty of Pure and Applied Mathematics, Wrocław University of Science and Technology, 50-370 Wrocław, Poland

Interests: applied probability; stochastic processes; risk theory; mathematical finance; stochastic control and optimization; extreme value theory; options; queueing and population dynamics; equity-linked products; longevity risk



Dr. Pierre Patie

<u>Website (http://www.orie.cornell.edu/people/profile.cfm?netid=pp396)</u> <u>SciProfiles (https://sciprofiles.com/profile/1691534)</u> Editorial Board Member

Operations Research and Information Engineering, Cornell University, 220 Rhodes Hall, Ithaca, NY 14853, USA

Interests: insurance mathematics; ruin theory; path dependent options; point processes

Special Issues, Collections and Topics in MDPI journals

Prof. Dr. Gareth W. Peters

<u>Website (https://iris.ucl.ac.uk/iris/browse/profile?upi=GWPET42)</u> <u>SciProfiles (https://sciprofiles.com/profile/160355)</u> Editorial Board Member

Department of Statistical Science, University College London (UCL), London, UK

Interests: quantitative risk management; insurance; computational statistics; machine learning; data analytics; econometrics; computational finance Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Huyên Pham

Wessite (https://sites.google.com/site/phamxuanhuyen/) SciProfiles (https://sciprofiles.com/profile/567964)

Editorial Board Member

Labertoire de Probabilités, Statistique et Modèlisation (LPSM); Université de Paris, Paris, France Interests: stochastic control; mean-field control/game; risk management; mathematical finance; machine learning techniques in finance

Dr. Dan Pirjol

Website (https://faculty.stevens.edu/dpirjol) SciProfiles (https://sciprofiles.com/profile/1409502)

Editorial Board Member

School of Business, Stevens Institute of Technology, Hoboken, NJ 07030, USA

Interests: derivatives pricing; risk management; applied mathematics

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Michael R. Powers

Website (http://www.sem.tsinghua.edu.cn/en/powers)

Editorial Board Member

Zurich Group Chair, Risk Mathematics, School of Economics and Management Tsinghua University, Room 386C, Weilun Building, Beijing 100084, China

Interests: financial regulation and public policy; game theory in risk and insurance; mathematical models in enterprise risk management; tax treatment of risk transfers; cultural attitudes and risk finance



Prof. Dr. Salvador Cruz Rambaud

<u>Website (https://cms.ual.es/UAL/personas/persona.htm?id=505550525048525775)</u> <u>SciProfiles (https://sciprofiles.com/profile/321432)</u> Editorial Board Member

Department of Economics and Business, Universidad de Almería, La Cañada de San Urbano, s/n, 04120 Almería, Spain

Interests: behavioral finance; intertemporal choice; microfinance; mathematical finance

Special Issues, Collections and Topics in MDPI journals

Dr. Luca Regis

Website (https://sites.google.com/view/lucaregis) SciProfiles (https://sciprofiles.com/profile/167109)

Editorial Board Member

Department of Socio-Economic and Mathematical-Statistical Sciences, University of Torino, Corso Unione Sovietica 218/bis, 10134 Torino, Italy Interests: actuarial mathematics; insurance; risk management; longevity risk; asset-liability management; financial mathematics; corporate finance Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Ramona Rupeika-Apoga

<u>Website (https://ej.uz/R_Rupeika_Apoga)</u> <u>SciProfiles (https://sciprofiles.com/profile/1189639)</u> Editorial Board Member

Faculty of Business, Management and Economics, University of Latvia, LV-1586 Riga, Latvia

Interests: FinTech; alternative financing; international finance; banking

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. José María Sarabia

<u>Website (https://www.cunef.edu/claustro/sarabia-alegria-jose-maria/)</u> <u>SciProfiles (https://sciprofiles.com/profile/1295755)</u> Editorial Board Member

Department of Quantitative Methods, CUNEF University, Leonardo Prieto Castro 2, 28040 Madrid, Spain

Interests: distribution theory; economic inequality; risk analysis; informetrics; multivariate analysis

Special Issues, Collections and Topics in MDPI journals

Prof. Michael Sherris

<u>Website (https://research.unsw.edu.au/people/professor-michael-sherris)</u> <u>SciProfiles (https://sciprofiles.com/profile/237271)</u> Editorial Board Member

Risk and Actuarial Studies and Centre of Excellence in Population Ageing Research (CEPAR), UNSW Business School, University of New South Wales, Sydney, NSW 2052, Australia

Interests: longevity risk; mortality models; long-term care insurance; retirement financing; quantitative risk management Special studes, Collections and Topics in MDPI journals



<u>Ç X (/toggle_desktop_layout_cookie)</u> Q ≡

Prof. Dr. Pavel Shevchenko

<u>Website (https://www.mq.edu.au/research/research-centres-groups-and-facilities/prosperous-economies/centres/centre-for-risk-analytics/our-people/pavel-shevchenko)</u> <u>SciProfiles (https://sciprofiles.com/profile/156408)</u>

Editorial Board Member

Department of Actuarial Studies and Business Analytics, Macquarie University, Sydney, NSW 2109, Australia

Interests: modelling extreme events; dependence modelling; state-space models; Monte Carlo methods; optimal stochastic control; machine learning methods

Special Issues, Collections and Topics in MDPI journals

Prof. Dr. Ajay Subramanian

<u>Website (https://robinson.gsu.edu/profile/ajay-subramanian/)</u> <u>SciProfiles (https://sciprofiles.com/profile/545261)</u> Editorial Board Member

Department of Risk Management & Insurance, Robinson College of Business, Georgia State University, Atlanta, GA, USA **Interests:** financial economics; insurance economics; mathematical finance; microeconomic theory; industrial organization <u>Special Issues, Collections and Topics in MDPI journals</u>

Prof. Dr. Anatoliy Swishchuk

Website (https://people.ucalgary.ca/~aswish/) SciProfiles (https://sciprofiles.com/profile/140249)

Editorial Board Member

Department of Mathematics and Statistics, University of Calgary, Calgary, AB T2N 1N4, Canada

Interests: mathematical finance; energy finance; stochastic modelling; risk theory; random evolutions and their applications; modeling high-frequency and algorithmic trading; deep and machine learning in quantitative finance

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Alexander Szimayer

<u>Website (https://www.wiso.uni-hamburg.de/fachbereich-sozoek/professuren/szimayer/team/szimayer-alexander.html)</u> <u>SciProfiles (https://sciprofiles.com/profile/222140)</u>

Editorial Board Member

Faculty of Economics and Social Sciences, University of Hamburg, 20146 Hamburg, Germany

Interests: financial econometrics; mathematical finance; economics of risk and insurance

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Josef Teichmann

<u>Website (https://people.math.ethz.ch/~jteichma/)</u> <u>SciProfiles (https://sciprofiles.com/profile/1415390)</u> Editorial Board Member

ETH Zürich, 8092 Zürich, Switzerland

Interests: mathematical finance; machine learning in finance; rough analysis



Prof. Dr. Weidong Tian

<u>Website (http://belkcollegeofbusiness.uncc.edu/wtian1/)</u> <u>SciProfiles (https://sciprofiles.com/profile/54438)</u> Editorial Board Member

Department of Finance, University of North Carolina at Charlotte, 9201 University City Blvd., Charlotte, NC 28223-0001, USA Interests: asset pricing; risk management; Knightian uncertainty; derivative and insurance market

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Emiliano A. Valdez

Department of Mathematics, University of Connecticut, Storrs, CT, USA

Interest: work and dependencies; elliptical distributions and their applications; managing post-retirement assets; longevity risks and annuitization; risk measures and capital requirements; applications of financial economics in actuarial science; competing risks models; survival ana terms and their applications of financial economics in actuarial science; competing risks models; survival ana terms and terms are terms and terms and terms and terms and terms and terms and terms are terms and terms and terms and terms are terms and terms and terms and terms are terms and terms and terms are terms are terms are terms and terms are terms are

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Frédéric Vrins

<u>Website (https://uclouvain.be/fr/repertoires/frederic.vrins)</u> <u>SciProfiles (https://sciprofiles.com/profile/369690)</u> Editorial Board Member

Louvain Finance Center & CORE, Université catholique de Louvain, 1348 Louvain-la-Neuve, Belgium

Interests: credit risk modelling; counterparty risk; stochastic calculus; information theory

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Joël Wagner

<u>Website (http://people.unil.ch/joelwagner/joel-wagner/)</u> <u>SciProfiles (https://sciprofiles.com/profile/356279)</u> Editorial Board Member

- 1. Department of Actuarial Science, Faculty of Business and Economics (HEC Lausanne), University of Lausanne, 1015 Lausanne, Switzerland
- 2. Swiss Finance Institute, University of Lausanne, Lausanne, Switzerland

Interests: actuarial science; insurance economics; risk management; strategic management



Prof. Dr. Alex Weissensteiner

<u>Website (https://sites.google.com/site/alweissensteiner/home)</u> <u>SciProfiles (https://sciprofiles.com/profile/168414)</u> Editorial Board Member

School of Economics and Management, Free University of Bozen-Bolzano, Bolzano, Italy

Interests: asset allocation; financial; risk management; financial engineering

Special Issues, Collections and Topics in MDPI journals



Prof. Dr. Wing-Keung Wong

Website (http://dns2.asia.edu.tw/~wong/) SciProfiles (https://sciprofiles.com/profile/278080)

Editorial Board Member

- 1. Department of Finance, College of Management, Asia University, Wufeng, Taichung 407705, Taiwan
- 2. Department of Medical Research, China Medical University Hospital, Taichung City, Taiwan
- 3. Department of Economics and Finance, The Hang Seng University of Hong Kong, Hong Kong, China

Interests: behavioral models; mathematical modeling; econometrics; energy economics; equity analysis; investment theory; risk management;

behavioral economics; operational research; decision theory; environmental economics; public health; time series analysis; forecasting <u>Special Issues, Collections and Topics in MDPI journals</u>



Prof. Dr. Sheung Chi Phillip Yam

Website (http://www.sta.cuhk.edu.hk/scpy)

Editorial Board Member

Department of Statistics, The Chinese University of Hong Kong, Shatin, N.T., Hong Kong, China Interests: actuarial science; applied mathematics; mathematical finance; probability theory and stochastic analysis; statistical theory and applications

Dr. Kazutoshi Yamazaki

Website (https://sites.google.com/site/kyamazak/) SciProfiles (https://sciprofiles.com/profile/254831)

Editorial Board Member

Department of Mathematics, Faculty of Engineering Science, Kansai University, 3-3-35 Yamate-cho, Suita-shi, Osaka 564-8680, Japan **Interests:** stochastic processes; optimal stopping; optimal stochastic control; risk theory; applied probability; actuarial science



Prof. Dr. Hailiang Yang

Wessie http://www.saasweb.hku.hk/staff/hlyang/) SciProfiles (https://sciprofiles.com/profile/403613)

Editorial Board Member

Department of Statistics and Actuarial Science, The University of Hong Kong, Pokfulam, Hong Kong $\sum (ltoggle_desktop_layout_cookie) \ Q \equiv$ Interests: equity-linked insurance products; insurance risk models; optimal control and applications to finance and insurance; option pricing Special Issues, Collections and Topics in MDPI journals



Dr. Ricardas Zitikis <u>Website (http://fisher.stats.uwo.ca/faculty/zitikis/main.htm)</u>. <u>SciProfiles (https://sciprofiles.com/profile/42706)</u>. *Editorial Board Member* Department of Statistical and Actuarial Sciences, Western University, London, ON N6A 5B7, Canada Interests: risk analysis and management; risk measures; econometrics <u>Special Issues, Collections and Topics in MDPI journals</u> <u>Risks (/journal/risks)</u>. EISSN 2227-9091, Published by MDPI <u>RSS (/rss/journal/risks)</u>. <u>Content Alert (/journal/risks/toc-alert)</u>.

Further Information

Article Processing Charges (/apc) Pay an Invoice (/about/payment) Open Access Policy (/openaccess) Contact MDPI (/about/contact) Jobs at MDPI (https://careers.mdpi.com)

Guidelines

For Authors (/authors) For Reviewers (/reviewers) For Editors (/editors) For Librarians (/librarians) For Publishers (/publishing_services) For Societies (/societies) For Conference Organizers (/conference_organizers) **MDPI** Initiatives Sciforum (https://sciforum.net) MDPI Books (https://www.mdpi.com/books) Preprints.org (https://www.preprints.org) Scilit (https://www.scilit.net) SciProfiles (https://sciprofiles.com) Encyclopedia (https://encyclopedia.pub) JAMS (https://jams.pub) Proceedings Series (/about/proceedings)

Follow MDPI

LinkedIn (https://www.linkedin.com/company/mdpi) Facebook (https://www.facebook.com/MDPIOpenAccessPublishing) Twitter (https://twitter.com/MDPIOpenAccess)

Subscribe to receive issue release notifications and newsletters from MDPI journals

Select options

Priter your email address	
Cubeerike	

© 1996-2023 MDPI (Basel, Switzerland) unless otherwise stated

Disclaimer Terms and Conditions (/about/terms-and-conditions) Privacy Policy (/about/privacy)

Vol. 3 (2023) (/2227-9091/11) Vol. 9 (2021) (/2227-9091/10) Vol. 9 (2021) (/2227-9091/9) Vol. 7 (2019) (/2227-9091/8) Vol. 7 (2019) (/2227-9091/7) Vol. 6 (2018) (/2227-9091/6) Vol. 5 (2017) (/2227-9091/6) Vol. 4 (2016) (/2227-9091/4) Vol. 3 (2015) (/2227-9091/4) Vol. 2 (2014) (/2227-9091/2) Vol. 1 (2013) (/2227-9091/1)

Risks — Indexing & Archiving

Risks is covered by the following databases and archives:

Indexing & Abstracting Services

- BibCnrs (https://bib.cnrs.fr/home/)
- CNKI (https://oversea.cnki.net/kns/AdvSearch)
- Digital Science (https://www.digital-science.com/)
- DOAJ (http://www.doaj.org/)
- EBSCO (https://www.ebsco.com/products/ebscohost-research-platform)
- EconBiz (https://www.econbiz.de/)
- EconLit (https://www.aeaweb.org/econlit/)
- Elsevier Databases (https://www.elsevier.com/solutions)
 - Scopus (https://www.scopus.com/home.uri)
- Gale (https://www.gale.com/intl/c/academic-onefile)
- OpenAIRE (https://explore.openaire.eu/)
- ProQuest (https://www.proquest.com/)
- RePEc (http://www.repec.org/)
 - EconPapers (https://econpapers.repec.org/scripts/search.pf)
 - IDEAS (http://ideas.repec.org/)
- SafetyLit (http://www.safetylit.org/index.htm)
- Web of Science (https://www.webofscience.com/wos/woscc/basic-search)
 - ESCI (https://www.webofscience.com/wos/woscc/basic-search)

Journal Ranking Lists

- ABDC Journal Quality List (Australian Business Deans Council) (https://abdc.edu.au/research/abdc-journal-quality-list/)
- <u>ANVUR (https://www.anvur.it/attivita/classificazione-delle-riviste/classificazione-delle-riviste-ai-fini-dellabilitazione-scientificanazionale/elenchi-di-riviste-scientifiche-e-di-classe-a/)</u>
- Journal Citation Reports (JCR) (https://jcr.clarivate.com)
- Norwegian Register for Scientific Journals, Series and Publishers (https://kanalregister.hkdir.no/publiseringskanaler/Forside.action?
 request_locale=en)
- Publication Forum JUFO (Federation of Finnish Learned Societies) (https://jfp.csc.fi/en/)
- <u>SCImago Journal & Country Rank (http://www.scimagojr.com/index.php)</u>

Journal Directories

- Electronic Journals Library (EZB) (http://rzblx1.uni-regensburg.de/ezeit/search.phtml?bibid=AAAAA&colors=7&lang=en)
- JournalGuide (https://www.journalguide.com/)
- Sherpa Romeo (https://v2.sherpa.ac.uk/romeo/)

Repository delivery services

- DeepGreen (https://info.oa-deepgreen.de/en/)
- Jisc (https://www.jisc.ac.uk/)

Digital Preservation

- CLOCKSS (https://clockss.org/)
- Swiss National Library (Helveticat) (https://www.e-helvetica.nb.admin.ch/)

Content Aggregators

- <u>Google Scholar (https://scholar.google.com/)</u>
- Scilit (https://app.scilit.net/)
- WorldCat (OCLC) (https://www.worldcat.org/)

<u>Risks (/journal/risks)</u>, EISSN 2227-9091, Published by MDPI <u>RSS (/rss/journal/risks)</u> <u>Content Alert (/journal/risks/toc-alert)</u>

Further Information

Article Processing Charges (/apc) Pay an Invoice (/about/payment) Open Access Policy (/openaccess) Contact MDPI (/about/contact) Jobs at MDPI (https://careers.mdpi.com)

Guidelines

For Authors (/authors) For Reviewers (/reviewers) For Editors (/editors) For Librarians (/librarians) For Publishers (/publishing_services) For Societies (/societies) For Societies (/societies) For Conference Organizers (/conference_organizers) MDPI Initiatives Sciforum (https://sciforum.net) MDPI Books (https://www.mdpi.com/books) Preprints.org (https://www.preprints.org) Scilit (https://www.scilit.net) SciProfiles (https://sciprofiles.com) Encyclopedia (https://encyclopedia.pub)

Follow MDPI

JAMS (https://jams.pub)

Proceedings Series (/about/proceedings)

<u>LinkedIn (https://www.linkedin.com/company/mdpi)</u> <u>Facebook (https://www.facebook.com/MDPIOpenAccessPublishing)</u> <u>Twitter (https://twitter.com/MDPIOpenAccess)</u>





Article

Rational versus Irrational Behavior of Indonesian Cryptocurrency Owners in Making Investment Decision

Elisa Tjondro, Saarce Elsye Hatane, Retnaningtyas Widuri and Josua Tarigan





https://doi.org/10.3390/risks11010017



Article



Rational versus Irrational Behavior of Indonesian Cryptocurrency Owners in Making Investment Decision

Elisa Tjondro * D, Saarce Elsye Hatane D, Retnaningtyas Widuri D and Josua Tarigan D

Department of Accounting, School of Business and Management, Petra Christian University, Siwalankerto 121-131, Surabaya 60236, Indonesia

* Correspondence: elisatjondro@petra.ac.id

Abstract: The purpose of this study is to investigate the salient factors that influence Indonesian cryptocurrency owners in making their investment decision. This study employs intergroup bias, subjective norms, overborrowing, and spending control to explain cryptocurrency investment behavior. The questionnaire was collected from 309 respondents from the five largest internet user areas: Jakarta, Surabaya, Bandung, Semarang, and Medan. This study executes the research framework using binary logistic regression. The results reveal that intergroup bias and overborrowing are the most impulsive factors contributing to the cryptocurrency investment decisions over the past year. Furthermore, after November 2021, Indonesian crypto owners are more irrational in a bearish period since their investment decisions are driven by their desire to be accepted in the social group. Moreover, when they have overindebtedness, instead of solving their debt problems, they prefer to spend their money on cryptocurrency investments. The subjective norms' influencers suggest that crypto owners not invest when the cryptocurrency price is sharply declining. The findings contribute to the dual-systems perspective and social contagion theories, enriching the empirical study regarding investment behavior.

Keywords: cryptocurrency; intergroup bias; subjective norms; self-control; overborrowing; spending control; survey

1. Introduction

Cryptocurrency investors frequently act irrationally in making investment decisions. This study explores how intergroup bias, subjective norms, and self-control factors influence the rationality of investment decisions. Empirical evidence has shown that investors do not always act rationally (Ahmad and Wu 2022), including in the cryptocurrency market. Previous research has been conducted on the behavioral bias in the equity market (Kumari et al. 2020; Ahmad and Wu 2022; Ahmad 2022; Ahmed et al. 2022; Lei and Salazar 2022; Liang et al. 2022), commercial real estate market (Kinatta et al. 2022), and cryptocurrency market (Ryu and Ko 2019). Ryu and Ko (2019) have conducted exceptional research on cryptocurrency investment decisions, which shows that strong impulses and weak self-control impact speculative bitcoin investments. The study of behavioral bias can help in understanding individual investors from different environments, resulting in discrete investment decisions. An individual has a common tendency to imitate, refer, and observe other behavior, specifically in a declining or unstable market condition (Yu et al. 2018; Shah et al. 2019).

In this study, there are four types of behavioral bias: intergroup bias, subjective norms, overborrowing, and spending control. First, intergroup bias is a tendency to behave more positively and provide greater rewards for their group members than outside groups (De Dreu and Kret 2016; Fujino et al. 2020). Intergroup bias in this study is focused on bias originating from a secondary group of investors' social environment, which is identical with lower intimacy and a lower frequency and duration of interaction—for example,



Citation: Tjondro, Elisa, Saarce Elsye Hatane, Retnaningtyas Widuri, and Josua Tarigan. 2023. Rational versus Irrational Behavior of Indonesian Cryptocurrency Owners in Making Investment Decision. *Risks* 11: 17. https://doi.org/10.3390/ risks11010017

Academic Editor: Mogens Steffensen

Received: 21 November 2022 Revised: 27 December 2022 Accepted: 5 January 2023 Published: 11 January 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). religion-based groups or sports groups. Although group members have similar interests, the members' purpose is to build social networks, bridging and bonding capital (Lei and Salazar 2022) that can increase the members' income and wealth status (Zhang et al. 2018). This is consistent with the findings of Chan et al. (2022), which suggest that collectivist social values influence individual financial behavior due to a sense of solidarity in homogeneous communities. Furthermore, in a game task experiment, intergroup bias impacts individuals' tendency to invest more in their group than in outside groups (Fujino et al. 2020). Individual behavior that is more positive towards their group members potentially results in irrational investment decisions since the trust bias toward their group influences the investment decision. The social contagion theory supports this argument (Bakker et al. 2010). Second, in the theory of planned behavior (TPB), subjective norms refer to beliefs about the expectations from peers and the most important persons to an individual, which motivates the individual to fulfill these expectations. Subjective norms in this study focus on a primary group of investors' social environment: peers, the most important persons, and the price trend. Subjective norms are a significant determinant influencing investment decisions—for instance, adopting and using technology (Ajzen 1991). Third, overborrowing reflects financial behavior related to high credit interest or excessive loans (Kawamura et al. 2021). Overborrowing is frequently associated with the impulsive behavior of buying or investing without thinking about the future. Investors associated with high overborrowing behaviors have a propensity for investing in cryptocurrency, though in a high uncertainty period, they support an irrational investment decision. Finally, spending control bias is a compulsive buying behavior associated with unstable, self-inconsistent, and negative emotions and perceptions of oneself (Liu and Zhang 2021). Weak spending control behavior can generate irrational investment decisions. However, studies on behavior bias in cryptocurrency investment decisions, particularly intergroup bias, subjective norms, and self-control bias, are still limited, and this motivates this study. To fill this gap, this study aims to expand on Ryu and Ko (2019) by examining whether intergroup bias, subjective norms, and self-control bias influence the investment decisions regarding whether to invest or not in the cryptocurrency market.

From 2021 until the third quarter of 2022, cryptocurrency markets faced enormous challenges, with a very significant decline in market value, even though there have been several small surges in the past few weeks. This study collected a survey from crypto owners regarding their decisions during that period. The first half of 2022 was a terrible period for the cryptocurrency market. Bitcoin and Ethereum, the two largest cryptocurrencies, declined by more than 50 percent from their highs in November 2021 (Gailey and Haar 2022). Based on Figure 1, the Bitcoin market, compared to the Indonesian Rupiah, had decreased by around 67% since its highest position on 8 November 2021 (963 million to 314 million on 20 August 2022, when the data collection was performed). The bearish market has the potential to influence investors' perceptions as market participants as well as the social environment in which investors interact. Then, it has an impact on rational or irrational investors' behavior in cryptocurrency investment decisions. Uncertainty conditions generally lead to various positive or negative attitudes in the social environment that can influence investors' investment decisions. Investors who decided to invest or not invest in cryptocurrencies over the past year show that the dual-system perspective, which is a reflexive and reflective system, runs in harmony in the decision-making process. The reflexive system is fast, impulsive, automatic, and unconscious, whereas the reflective system is slow, controlled, conscious, and analytical (Ryu and Ko 2019). The factors of intergroup bias, subjective norms, overborrowing, and spending control bias can trigger the dual-system perspective, resulting in rational or irrational behaviors in the investment decisions.



Figure 1. Bitcoin to Indonesian Rupiah from November 2021 to November 2022. Source: Google Finance (2022).

Positive or negative attitudes towards cryptocurrency investment originate from the investors' social environment and create a more significant gap when the market is in a declining condition. The cryptocurrency market has unique characteristics that are different from those of conventional markets—for example, stocks and property markets. Cryptocurrencies provide a new alternative investment. Individuals believe that digital money is the money of the future (Bhatt 2022), and the number of users is increasing progressively. However, cryptocurrencies are risky speculative investments despite their inherent digital future potential. Most Southeast Asian countries, including Indonesia, Malaysia, and Vietnam, consider cryptocurrency illegal as a medium of exchange but legal as an investment or commodity. In addition, Thailand has just started to tighten the regulation of cryptocurrencies (Cointelegraph 2022). Therefore, it is critical for investors and potential investors to understand the applicable regulations and make decisions with a complete understanding of the potential risks.

Individual investors from the same geographic area were more likely to adopt biased behavior than cross-country investors (Choi 2016). Indonesia has seen a 280 percent growth in the number of crypto investors since 2020, from 1.5 million to 4.2 million individuals, with a daily trading volume reaching USD 117.4 million (Blockchain Association of Indonesia 2022). A study by Gemini (2022) entitled "Global State of Crypto Report" found that 41 percent of Indonesians between the ages of 18 and 75 with an income of more than USD 14,000 per year own cryptocurrencies. The research also found that 61 percent of Indonesian respondents agree that crypto is the future of money, which is the highest rate in the Asia Pacific (Gemini 2022). This study uses data from the Indonesian cryptocurrency market for three main reasons. First, there has been an acceleration of digital economic growth in Indonesia after the COVID-19 pandemic. As the largest economy in Southeast Asia, Indonesia has shown a significant increase in the value of the digital industry, from USD 41 billion in 2019 to USD 77 billion in 2022. It is driven primarily by e-commerce (Google, Temasek, Bain & Company 2022). The digital financial services increase is dominated by digital investment, which increased to 31% CAGR (Compound Annual Growth Rate) in 2022 and will increase to an estimated 74% CAGR in 2025 (Google, Temasek, Bain & Company 2022). These data show the significant potential of Indonesia's digital investment in Southeast Asia. Second, the number of individual Indonesian investors investing in cryptocurrency is greater than those investing in stocks in 2022. As of June 2022, the number of cryptocurrency investors was 15.1 million versus 9.1 million stock investors, despite the fact crypto investment is still relatively new in Indonesia (CNBC Indonesia 2022). Third, cryptocurrency investors in Asia are dominated by the young generation (Fujiki 2020, 2021; Santoso and Modjo 2022), so they fit the Indonesian demographic profile. Based on data for 2022, 78 percent of Indonesian crypto owners are between the age of 18 and 44 (TripleA 2022). Therefore, the Indonesian market provides a unique setting for researchers to analyze the influence of individual bias behavior on cryptocurrency investment decisions.

The findings of this study provide novel evidence supporting the dual-system perspective and contagion theory by emphasizing the importance of understanding the influence of intergroup bias, subjective norms, and self-control bias on investors' rational or irrational behavior in the decision-making process. Recent studies investigate the effect of subjective norms and self-control (Ryu and Ko 2019), financial literacy and investment experience (Zhao and Zhang 2021; Fujiki 2021), and attitude towards and trust in cryptocurrency investment decisions (Stix 2021). This research is different from these studies in the following ways. First, although the Ryu and Ko (2019) study was conducted during the declining market of cryptocurrencies, the Ryu and Ko (2019) study did not discuss the intergroup bias factor and did not analyze which factors determine investment decisions in cryptocurrencies. Second, although Zhao and Zhang (2021), Fujiki (2021), and Stix (2021) found that several factors were proven to influence crypto owners' investment decisions, their studies did not address intergroup bias, subjective norms, and self-control bias as factors influencing the decisions. Thus, this study is novel, since this study demonstrates that intergroup bias and subjective norms result in different stimuli for crypto owners' investment decisions. Intergroup bias contributes to the decision to invest in the market despite the declining conditions. On the other hand, subjective norms contribute to the decision to not invest in cryptocurrencies when market conditions experience a significant decline. This study also finds that overborrowing can result in the irrational behavior of investors who keep investing in cryptocurrency in declining conditions.

This study contributes to the cryptocurrency literature in the following ways. First, this study contributes to the development of cryptocurrency literature in Asia, which is synonymous with a collectivist culture that is vulnerable to the contagion effect of investment behavior. It provides empirical evidence supporting dual-system and social contagion theories by identifying intergroup bias, subjective norms, and overborrowing bias as the impulsive factors contributing to the cryptocurrency investment decisions. Second, an analysis of individual investors' biased behavior is performed in an extreme declining period which is still limited. During periods of significant market decline, the risk associated with cryptocurrency investments for owners with vulnerable risks, such as contagion risk and financial risk, increases. Therefore, irrational investors are more inclined to invest in cryptocurrency during adverse periods. Finally, this study enhances the behavioral finance literature on the rational and irrational behavior of crypto owners in making investment decisions by providing evidence of the effect of intergroup bias, subjective norms, and overborrowing bias on cryptocurrency investment decisions.

The discussion of this study is divided into several sections. Section 2 discusses the literature review and hypothesis development. Section 3 describes the research methodology, including the sample selection and analysis model. Section 4 shows the results of the statistical tests, the interpretation of the results, and theoretical and practical implications. Finally, Section 5 describes the conclusions, the limitations of the study, and potential future research.

2. Literature Review

2.1. Intergroup Bias and Subjective Norms in Cryptocurrencies Investment Decisions

Behavioral research of individual crypto owners, especially in emerging markets, is an interesting topic and has broad future potential. Kumar et al. (2022), who conducted a bibliometric study in the field of behavioral finance, suggest that additional research is needed to understand the factors that influence investors' behavior in the markets. According to Kumar et al. (2022), individual decision makers differ fundamentally, contributing to differences in financial behavior in investment decision making. The suggestion from Da Gama Silva et al. (2019) is to analyze bias behavior when the cryptocurrency market is in a sharp decline, since it has the potential to provide new findings for the development of the literature.

This study discusses biased behavior with two focuses: intergroup bias and subjective norms. First, intergroup bias is the tendency for members of one group to behave more positively and provide greater rewards than those outside the group (De Dreu and Kret 2016; Fujino et al. 2020). Collectivist culture has a strong positive influence on financial

behavior (Chan et al. 2022). The impact of intergroup bias on the investment market is more destructive than the subjective norms because it involves more irrational and illogical thinking and blindly imitates the actions of others because of psychological and emotional factors. The need to be recognized as part of a social group can also lead to biased behavior in investment decisions. Second, subjective norms are beliefs about the expectations or important references of others that motivate investors to meet these expectations (Ajzen 1991)—for example, expectations from peers, the most important persons, and references to market trends. Crypto owners are inclined to follow advice from the closest social environment, including peers, the most important persons in making investment decisions, and price trends. Crypto investors rely on peers to reduce their potential risk due to wrong investment decisions. Bias investors try to match the investment performance of peers by relying on others' investment decisions. The two types of biased behaviors have different motivations and produce different behaviors in investment decisions.

2.2. Hypothesis Development

2.2.1. Intergroup Bias and Contagion Effect in Collectivist Culture

Intergroup bias is a bias behavior of decision making caused by the contagion effect from a secondary social group. The contagion effect arises because investors have a high level of trust (Bakker et al. 2010) towards other members of the social club through regular interaction. The perspective of social contagion theory is relevant to investment decisions since individuals tend to adopt similar behavior when they trust the information provided by members of their social network (Westaby et al. 2014). The three characteristics of the social contagion effect that influence intergroup bias are being aware of the knowledge that others have, appreciating what other persons know, and gaining access to one's thinking patterns (Borgatti and Cross 2003). The convergence of attitudes and beliefs depends on exposure to information obtained when communicating between social networks or groups (Peters et al. 2017). The interaction between members of a social group or social network, as bridging and bonding capital (Lei and Salazar 2022), can increase the members' wealth status. Furthermore, biased investors obtain financial knowledge, resources, and business opportunities from their groups, which in turn increase their income (Zhang et al. 2018).

Areas inhabited by various ethnic groups are one of the causes of massive contagion effect behavior (Chan et al. 2022). Concern over one's immediate ethnic survival leads to solidarity in genetically homogeneous communities, which is conducive to developing collectivist cultures (Chan et al. 2022). Cultures based on certain groups or ethnicities, such as beliefs, norms, and social values, tend to remain or not easily change over a long time (Gorodnichenko and Roland 2017). The existence of various ethnic groups in Indonesia is one of the motivating factors for research on intergroup bias.

Hypothesis 1 (H1). *Intergroup bias has contributed to cryptocurrency investment decisions over the past year.*

2.2.2. Subjective Norms and Cryptocurrency Investment Decisions

The subjective norm in this study focuses on the influence of the closest social environment, which are peers and the most important persons, and market trends in the cryptocurrency market. During an adverse period, influencers of subjective norms tend to suggest not to invest in cryptocurrencies. They prefer to manage their risk exposure since cryptocurrency is a speculative investment (Ryu and Ko 2019). Long-term goals, rational, and analytical, are the drivers of the influencers of subjective norms. The primary social group of investors—for example, peers, the most important persons—contribute to the decision to avoid investing in cryptocurrency during a terrible period. Ouimet and Geoffrey (2020) found that peers from the same employer firm influence individual financial decisions.

In contrast, subjective norms may also result in investors' impaired technical knowledge and reasoning abilities, causing errors in judgment. Consequently, investors make irrational decisions, which can adversely affect their returns (Ahmad and Wu 2022). Based on the arguments that have been explained, the research hypothesis is

Hypothesis 2 (H2). *Subjective norms have contributed to cryptocurrency investment decisions over the past year.*

2.2.3. Self-Control Behavior and Cryptocurrency Investment Decisions

Self-control is the ability to regulate emotions and behavior and inhibit individual impulses to achieve long-term results (Sekścińska et al. 2021). This study focuses on two dimensions of self-control: overborrowing and spending control. Previous studies that discussed financial behavior focused more on linking self-control with financial risk-taking or gambling risks, but issues related to investment choice were mostly ignored (Sekścińska et al. 2021). This study argues that overborrowing and spending controls contribute to the cryptocurrency investment decisions.

Overborrowing reflects financial behavior that is synonymous with high credit interest or excessive loans (Kawamura et al. 2021). Overborrowing is often associated with impulsive behavior, whereas, in decisions to buy or invest, individuals act without thinking about the future. Gathergood (2012) showed that individuals who act impulsively tend to use various types of credit, including consumer credit, which makes them more vulnerable to financial risk. This is in line with the studies by Friehe and Schildberg-Hörisch (2017) and Kocher et al. (2019), who found that low self-control increases risk-taking in investment decisions. Investors with a high level of debt tend to raise their risk exposure irrationally and invest in high-risk and speculative investments based on their emotions.

The other type of self-control is spending control. Low spending control is when individuals engage in impulsive consumer behavior or compulsive buying (Neuner et al. 2005). Liu and Zhang (2021) found that compulsive buying was associated with unstable, self-inconsistent, and negative emotions and perceptions of oneself. Furthermore, compulsive buying is similar to individuals who focus on materialistic values as a strategy to alleviate anxiety in response to insecurity symptoms (Kasser and Ahuvia 2002). Cryptocurrency is a speculative investment with a high volatility and the potential for greater returns, thus providing an impulse for individuals with low spending control. This study contends that low spending control contributed to the investment of cryptocurrencies in a highly uncertain period. In contrast, investors with high spending control will choose to refrain from investing in cryptocurrencies in a declining period.

Hypothesis 3a (H3a). Overborrowing behavior has contributed to cryptocurrency investment decisions over the past year.

Hypothesis 3b (H3b). *Spending control has contributed to cryptocurrency investment decisions over the past year.*

3. Methodology

3.1. Sample Selection and Questionnaire Study

The sample of this study is 309 respondents who are active crypto owners in Indonesia and are actively involved in social clubs. The period of distributing the questionnaire link lasted four weeks, from the beginning to the end of August 2022. The links were distributed to 997 respondents who are members of a pooled database of a credible and trusted survey service organization. Individuals who filled out the survey received a GoPay/OVO voucher of IDR 10,000, equivalent to USD 0.65. A total of 532 individuals did not have cryptocurrencies, so they were excluded from the sample. Forty respondents were disqualified for filling out the questionnaire incorrectly, twenty-five respondents did not complete filling out the questionnaire, and ninety-one respondents were not actively involved in any social club. This study uses several demographic criteria in the sample selection to avoid sample selection bias. First, the sample is an equal number of men and women. Second, the sample respondents came from five cities, with a balanced number of 80 respondents per city and a total of 400 respondents. Third, three age categories are the sample of this study, namely, 21–30 years, 31–41 years, and 41–50 years. Respondents who did not fall into these three categories were excluded from the sample. Finally, the respondents were actively involved in social clubs or clubs over the past year. A total of 91 respondents were not actively involved in social clubs or clubs; therefore, they were excluded from the sample.

This study conducted several stages of sample selection. First, this study accurately and precisely specified the population. The population of crypto owners in Indonesia is difficult to determine, and data are unavailable. Alternatively, this study uses the 2022 population of internet users provided by the Association of Indonesian Internet Service Providers (APJII). Previous studies have found that crypto owners are active internet users who use the internet at least once per week (Stix 2021). Second, the sampling frame phase was determined. A sampling frame is all the available elements of a population that has a chance of being selected for the survey (Dobosh 2018). The respondents in this study are verified members of a pooled database provided by a trusted surveyor service organization. The individual targets are crypto owners who have a basic understanding and are actively involved in social clubs. The respondent data based on the level of crypto knowledge indicated that as many as 67.6 percent have a basic knowledge of cryptocurrency and 25.9 percent have thorough knowledge. Third, the sampling technique was determined, and random probability sampling was used. Probability sampling is when all the elements of a sampling frame have an equal chance of being selected for the sample. The use of random sampling can reduce bias and increase the likelihood that the sample is representative (Dobosh 2018). The method of distributing the survey was conducted online for 1100 individuals who are verified members of a database of surveyor services organizations, so all individuals had the same opportunity to be selected as samples.

This study included a validation question in the form of simple mathematical addition to ensure that the respondents filled out the questionnaire consciously. If the respondent answers incorrectly, it is assumed that the respondent was not fully conscious when answering the question, and the respondent is disqualified. There are five sections to the questionnaire questions. The first is demographic information and the screening of currently active crypto owners. If the respondent does not have any cryptocurrencies, the respondent is disqualified. The second is information regarding cryptocurrency acquirement over the past year. The second part also asked about the subjective norms in cryptocurrency purchase decisions. The third section involves the questions of intergroup bias, beginning with the definition of a social club and the decision to participate actively in a social club. Respondents who answered that they had not been actively involved in a social club did not fill in this section and were excluded from the sample. In the questionnaire, our study provides examples of secondary social groups that are religious-based groups or sports clubs. This secondary group's members are the same people for a set period, so interactions occur regularly rather than just once or twice. Regular interaction with the social group is the main key to building trust between members, which can cause a contagion effect on investment behavior. In the fourth section, the respondents were asked about the overborrowing experienced over the past year. Finally, there was the question of overborrowing and spending control.

The period of crypto ownership is the past year. This study uses binary logistic regression and divides the dependent variable into two groups of crypto owners. Respondents who obtained cryptocurrencies over the past year are assigned number one, while those who do not meet the criteria are assigned number two. A year cap was chosen to limit the current motivation that causes respondents to buy cryptocurrencies. Stix (2021) stated that a more extended crypto buying period could introduce research bias, as motivations and influencing factors could potentially differ from those of the study population. In addition, when the survey was conducted, the cryptocurrency market conditions were decreasing. Da Gama Silva et al. (2019) also suggests analyzing biased behavior when the

cryptocurrency markets are in a sharp decline because it has the potential to provide new findings for the development of the literature.

This study uses binary logistic regression to analyze Indonesia's factors influencing crypto ownership over the past year, especially in five big cities concentrated in Java and Sumatra, Bandung, Semarang, Surabaya, Medan, and Jakarta. These five cities were chosen because the population of internet investors in these five cities represented 43.61% of the population of internet investors in Indonesia, with an average internet penetration ratio per province of 78.98% (APJII 2022). The internet investors per province and provincial capital are shown in Table 1.

Contribution to **Internet Penetration** The Largest Internet Users Capital of the Indonesia's Total **Based on Provinces Ratio of Each Province** Province **Internet Users** West Java Bandung 14.74% 82.4% 10.93% 72.9% East Java Surabaya Central Java 10.36% 76.9% Semarang North Sumatra 4.34%79.3% Medan DKI Jakarta Jakarta 3.24% 83.4% Total contribution nationally 43.61% Average internet penetration 78.98% per province

Table 1. The largest internet users in Indonesia by province capital city.

Source: Association of Indonesian Internet Service Providers (APJII 2022).

The sampling technique uses random sampling, and the sample size is determined based on the number of variables, where ten observations are needed for each variable studied. Peduzzi et al. (1996) and Peng et al. (2002) use a minimum sample ratio of 10 to 1, with a minimum sample size of 100. The formula is n = 10k/p, in which n is the number of minimum samples, k is the number of predictors, and p is the smallest proportion of binary cases in the population. The minimum sample size for a four-predictor model is 167; thus, the sample size of 309 respondents meets the requirement. Data collection was carried out through surveys with an online distribution in five provincial capitals with the most significant internet investors in Indonesia.

There are several stages to preparing the instrument. First, the questionnaires from previous references were translated into Indonesian and modified according to the research objectives. Second, the survey instrument was assessed by two experts: Professors and practitioners in investment and accounting behavior. Third, a pilot project was held for 30 individuals not included in the research sample. Questionnaire questions that do not pass the validity and reliability test will not be used in the survey. Finally, the instrument was translated back into English for publication purposes.

3.2. Definition of Variables and Model Analysis

3.2.1. Dependent Variables

The dependent variable of this study is cryptocurrency investment over the past year (PYI). The dependent variable is operationalized by a dummy variable, given a score of 1 if yes or a score of 2 otherwise. This study modifies the measurement of PYI from Stix's (2021) study by focusing more on exploring the factors contributing to the decision to invest in cryptocurrency over the past year, especially during periods of extreme decline in cryptocurrency markets. The dependent variable and the indicators are presented in Table 2.

Construct	Indicators	Code		
	Dependent variable			
Past year investment	Invest in cryptocurrencies over the past year.	PYI		
	Independent variables			
	The reason to invest in cryptocurrencies is to be recognized in a social group.			
	The reason to invest in cryptocurrencies is to follow the action of other group members.			
Intergroup bias	The reason to invest in cryptocurrencies is believing that other members have more knowledge about cryptocurrencies.	IB		
	The reason to invest in cryptocurrencies is the better performance of other group members.			
	Invest in a cryptocurrency whose value is rising in the market.			
	Investment decisions are based on the actions of others.			
	The reason to invest in cryptocurrencies is to follow the same pattern of decisions as other investors.			
Subjective norms	The reason to invest in cryptocurrencies is that friends or coworkers believe that investing in cryptocurrencies is popular.	SN		
	The reason to invest in cryptocurrencies is that the most important persons to me also invest in cryptocurrencies.			
	The reason to invest in cryptocurrencies is that people around me are doing so.			
	How frequently have you used consumer credit over the past year?			
Overborrowing	How frequently have you run out of money in your bank account over the past year?	OB		
0	How frequently have you had difficulty paying debts over the past year?			
	How frequently have you borrowed money at extremely high-interest rates over the past year?			
	When making spending decisions, I carefully consider my financial situation.			
Spending control	When making a cryptocurrency investment decision, I try to spend my money wisely.			
	When making a cryptocurrency investment decision, I try to put in only a little time or effort.			

Table 2. Definition of variables and indicators.

3.2.2. Predictor Variables

This study employs four predictor variables: intergroup bias, subjective norms, overborrowing, and spending control bias. The predictor variables use five-point Likert scales. Harpe (2015) stated that the response measured by five-point Likert scales is continuous data, so it is relevant to the independent variable in the logistic regression model. The intergroup bias modified the questions from the study by Kumari et al. (2020), and the subjective norm predictor used modified the questionnaire (Taylor and Todd 1995; Kumari et al. 2020; Kinatta et al. 2022). The instrument of overborrowing and spending control variables modified the research questionnaire (Eysenck and Eysenck 1978; Tangney June Price and Boone 2004; Kawamura et al. 2021; Sekita et al. 2022). The predictor variables and the indicators are presented in Table 2.

3.2.3. Demographic Variables

The demographic variables in this study include gender, city of residence, age, occupation, and activeness in social clubs. This study uses demographic variables as sample selection criteria to avoid sample selection bias. First, the respondents were divided into two gender groups (Male and Female) with equal numbers (Table 3). Second, the respondents came from five cities, with a balanced number of eighty respondents for each area (Table 3). Third, three age categories are the sample of this study, namely, 21–30 years, 31–41 years, and 41–50 years. Finally, the respondents were actively involved in social clubs or clubs over the past year. Several studies have found that demographic variables are associated with the ownership of cryptocurrencies. For example, Fujiki (2020) found that crypto owners in Japan are primarily male, are under 30 years old, have a high pretax income, work at private or public companies, are the main source of income from a business, and have a graduate school education level.

Table 3. Demographics of respondents.

Past Year Investment (PYI)	%	Gender	%	Age	%	Occupation	%	Area	%	
	Sample of 400 crypto owners									
Yes No	Yes30.0Male47.821–3049.8No70.0Female52.231–4037.341–5013.0		College student Private employee Business owner Full-time housewife Unemployment	4.3 76.5 15.3 3.3 0.8	Jabodetabek Surabaya Semarang Bandung Medan	20.0 20.0 20.0 20.0 20.0 20.0				
	A sample of 309 crypto owners actively involved in social clubs									
Yes No	35.9 64.1	Male Female	44.0 56.0	21–30 31–40 41–50	52.4 38.2 9.4	College student Private employee Business owner Full-time housewife Unemployment	4.5 74.8 17.2 2.9 0.6	Jabodetabek Surabaya Semarang Bandung Medan	23.3 16.2 24.9 23.9 11.7	

Research model:

$$\eta 1 = \eta \beta + \beta 1\xi 1 + \beta 2\xi 2 + \beta 3\xi 3 + \beta 4\xi 4 + \varepsilon \tag{1}$$

Information

 $\eta 1 = Past year investment (PYI)$

 $\eta\beta$ = Constant coefficient

- $\beta 1 \xi 1 =$ Intergroup bias (IB)
- $\beta 2\xi 2 =$ Subjective norms (SN)
- $\beta 3\xi 3 = Overborrowing (OB)$
- $\beta 4\xi 4$ = Spending self-control (SPC)
- ε = Error disturbance

4. Empirical Result

4.1. Demographic and Descriptive Statistics

In this study, 309 of the 400 crypto owners were actively involved in social clubs. Table 3 presents the demographics of respondents, including the percentage of past year investments (PYI), gender, age groups, type of occupation, and area. PYI describes respondents who invested in cryptocurrency over the past year as 35.90 percent of the total respondents. The remaining 64.10 percent are crypto owners who did not invest in cryptocurrency. The age group of respondents is between 21 and 50 years, whereas the age category of 21 to 30 years dominates by 52.4%. Most respondents work in the private sector (74.8 percent), followed by business owners at 17.2 percent in the next position. This study conducts the Fisher exact test between gender and the dependent variable. The results

show a significant (two-sided) Fisher exact test value of 0.153 (>0.05), confirming that the sample is free from gender bias problems.

Table 4 shows the descriptive statistics of the dependent variable of PYI using categorical data, whereas the answer "yes" is given the number 1, and 2 is given otherwise. Independent variables apply the mean score of the item indicators using a five-point Likert scale.

	Ν	Mean	STD	Min	Max	VIF
PYI	309	1.641	0.481	1	2	
IB	309	3.435	0.823	1.00	5.00	1.882
SN	309	3.306	0.754	1.20	4.80	1.859
OB	309	1.965	0.795	1.00	5.00	1.015
SPC	309	4.123	0.656	1.00	5.00	1.035

Table 4. Descriptive Statistics.

4.2. Hypothesis Result

The hypothesis testing begins with the determination of the validity and reliability of the indicators. The validity results using Pearson Correlation show coefficient values between 0.614 and 0.889 (r > 0.60) for each item indicator, so it can be concluded that item indicators can be used to measure the construct. The examination of reliability with Cronbach's alpha shows that a value greater than 0.60 can be interpreted as being of high reliability and as an acceptable index (Pallant 2001). The Cronbach's alpha values were 0.807, 0.807, 0.728, and 0.612 for IB, SN, OB, and SPC. The corrected item-total correlation ranged from 0.388 to 0.741, indicating good scales (Ferketich 1991). The Pearson correlation results in Table 5 display that the correlation coefficient between variables does not exceed 0.7. Thereby, it can be concluded that there is no strong correlation between variables, or it is at a moderate correlation level (Schober et al. 2018; McLeod 2022).

	РҮІ	SN	IB	OB	SPC
PYI	1				
SN	0.191 **	1			
IB	-0.028	0.675 **	1		
OB	-0.207 **	-0.008	0.025	1	
SPC	0.051	-0.004	0.104	-0.108	1

** significant at the 0.01 levels.

A binary logistic regression was employed to determine the impact of intergroup bias, subjective norms, and self-control derived from the factor analysis on the crypto investment decision. Table 6 shows the results of the logit model with the Wald test. The empirical result confirms that the intergroup bias (IB), subjective norms (SN), and overborrowing (OB) factors were significant ($\rho < 0.05$) predictors of the odds of PYI. In contrast, the spending control (SPC) factor was unconfirmed to predict the odds of PYI. Furthermore, the influencers in the group of subjective norms (SN) do not suggest that crypto owners invest during the heaviest period. On the contrary, intergroup bias (IB) and overborrowing (OB) have been predictors of cryptocurrency investment over the past year. This study found that overborrowing bias (OB) has a stronger predictive ability regarding investing in cryptocurrency than intergroup bias (IB). The exponential values of intergroup bias (IB) and overborrowing (OB) are 0.423 and 0.576, respectively, indicating that the ability to predict the odds ratio of PYI is greater in overborrowing (OB). The coefficient β 1 of intergroup bias (IB) reveals that the odds ratio of investing in cryptocurrencies over the past year decreases when the value of IB increases by one. The coefficient β 1 of the subjective norm (SN) is 1.204, and the exponential coefficient is 3.333, meaning that the odds ratio of investors who

did not invest in cryptocurrencies over the past year increases by 3.333 times as the value of SN increases by one when the other predictors are held constant.

	Dependent: Past Year Investment (PYI)							
	β	S.E.	Wald	Sig.	Exp(β)			
IB	-0.859	0.233	13.556	0.000	0.423			
SN	1.204	0.251	23.095	0.000	3.333			
OB	-0.551	0.162	11.577	0.001	0.576			
SPC	0.227	0.198	1.311	0.252	1.254			
Constant	-0.230	1.039	0.049	0.825	0.794			

Table 6. Coefficient of Predictor Factors.

The goodness-of-fit statistics assess the fit of the logit model to the actual outcomes (Peng et al. 2002). The omnibus test shows a significant model χ^2 (6) of 14.545 with a ρ -value of 0.000 ($\rho < 0.05$) for the PYI model. The -2 log likelihood (-2LL) estimate measures how well the estimated model fits with categorical data (Suthar et al. 2010). The value of -2LL for the model is 363,157. Hosmer and Lemeshow's (Hosmer et al. 1997) test result demonstrates a non-significant value of 0.069 ($\rho > 0.05$). Thereby, the model fit can be preserved.

A logit model is predicted accurately, including correctly predicting the outcome (Hosmer et al. 1997). Table 7 shows an ability to predict the PYI model of 73.463 percent. The ability to predict the crypto owners' decision not to invest in cryptocurrency (91.919 percent) is better than that predicting investment decisions over the past year (40.541 percent). The predictive ability of the model for cryptocurrency investment decisions (PYI = yes) is below 50 percent or weak. In other words, other factors not analyzed in the model influence investment decisions during the extreme declining period.

			Pree	dicted
		Past Year (P	Investment YI)	
Observed		Yes	No	Percentage Correct
Past Year Investment (PYI)	Yes	45	66	40.541
	No	16	182	91.919
Overall Percentage				73 463

Table 7. Predicted Results.

Several assumptions must be met in logistic regression. First, the linearity assumption uses box-tidwell transformation (Osborne 2017; Field 2018) to check for linearity between predictors and the logit. The results of the linearity test reveal that the box-tidwell transformation for the four independent variables is not significant in relation to the dependent variable, meaning that the linearity assumption is met. Second, the multicollinearity test in Table 4 shows a Variance Inflation Factor (VIF) value between 1.015 and 1.882 for the four independent variables.

4.3. Discussion

The results from this study show that intergroup bias (IB) and overborrowing (OB) behaviors are the most stimulating factors contributing to the cryptocurrency investment decision in an adverse market. Both factors contribute to irrational behavior in making an investment decision, specifically in the high-uncertainty conditions over the past year. Investors with the characteristics of having a high trust bias towards their social group and high overborrowing behavior had a tendency to invest in cryptocurrency over the past year when the price dropped by 67% from the highest point in November 2021 to

the end of August 2022, when the data collection was performed. However, this study provides evidence that subjective norms (SN) of the primary social environment caused crypto owners to refrain from investing in cryptocurrency over the past year.

Different types of social environments have distinctive effects on cryptocurrency investment decisions in a high-uncertainty market. The dual-system perspectives can explain these distinctive effects, which discuss reflexive and reflective perspectives (Ryu and Ko 2019). Cryptocurrency investment, as a speculative investment activity, emerges as a natural response to individual high- and low-impulse interactions (Ryu and Ko 2019). When impulsive and reflexive investors react most strongly, investors can make irrational decisions. Nevertheless, a reflective perspective encourages rational behavior. These two impulses go hand in hand. Investors can receive different impulses of investing or not investing in cryptocurrency simultaneously. The dual-system perspective, reflexive and reflective, does not occur in isolation but side-by-side in speculative cryptocurrency investments. Investors who decided to buy or not buy crypto over the past year show that the reflexive and reflective system runs in harmony in the decision-making process, whether influenced by intergroup bias or subjective norms. This is consistent with the findings of Da Gama Silva et al. (2019), who found that negative news in the cryptocurrency markets is related to behavior bias.

Intergroup bias from the secondary group of investors' social environments, such as religion-based groups and sports groups, contributes to irrational behavior when making an investment decision. There are two fundamental explanations related to intergroup bias and irrational behavior. First, investors who are actively involved as members in a social group tend to behave more positively, provide greater rewards, and have higher trust in the group members than outside groups, which is known as trust bias. The trust bias encourages investors to behave identically to their group members and make irrational investment decisions since they want to be recognized in the group. Furthermore, intergroup bias encourages investors to act fast, impulsive, automatically, and unconsciously in making investment decisions to obtain financial knowledge, resources, and business opportunities from their social groups, which can increase their income. Interaction between members of a social group or network, as bridging and bonding capital (Lei and Salazar 2022), can increase the members' income and wealth status (Zhang et al. 2018). Second, cryptocurrency investments provide different options because of their attractive characteristics, their high volatility, their higher average returns, the accessibility of weekend trading, and their low correlation with traditional assets. These characteristics are the advantages of investment diversification (Brière et al. 2015). Then, the social group, which generally prioritizes individual wealth status and exclusive networking, tends to stimulate the irrational behavior of crypto owners in making investment decisions.

Regarding the subjective norms, this study shows that the investors received a stimulus from the primary group of their social environment—for example, peers, the most important persons—and the price trends to not invest in cryptocurrency in the adverse period. Cryptocurrency is a speculative investment instead of a long-term investment (AFM 2022). Indonesian regulations state that crypto is illegal as a medium of exchange. However, it is allowed to be traded as a commodity (Jakarta Globe 2022), thereby expanding its function as a speculative investment. Previous studies also confirmed that Bitcoin is mainly used as a speculative asset rather than an alternative currency (Blau 2017; Baur et al. 2018). Thus, in a high-uncertainty condition, persons in the closest social environment of crypto owners tend to act cautious, slow, controlled, conscious, and analytically, exposing the reflective system. They try to convince crypto owners not to invest in cryptocurrency during the heaviest period.

Besides intergroup bias, the other impetus to invest in cryptocurrency in the declining market arises from overborrowing bias. The reason is that individuals with low self-control often act on a reflexive perspective, leading to high levels of unplanned (Friese and Hofmann 2009) and irrational behavior. This is consistent with the findings of Ryu and Ko (2019), who stated that strong impulses and weak self-control drive speculative investment

behavior in the cryptocurrency context. Easy access to fintech credit markets increases the risk of individuals falling into debt traps (Yue et al. 2022). Liu and Zhang (2021) explained that the easy access to online consumer credit became one of the causes of severe financial risk. The digital credit market trap is a challenge faced by crypto owners, who generally always come into contact with digital media.

The theoretical and practical implications of this study are described in several sections. First, there are minimal survey studies on biased behavior and cryptocurrency investment decisions, so this study enriches the literature on the irrational decisions of crypto owners, especially in Asia. Several crypto owners studies that are relevant to this study include those of Fujiki (2021) in Asia, Stix (2021) in Europe, and Zhao and Zhang (2021) in the USA. More specifically, studies with a sample of crypto owners in the Asia region have yet to receive much attention. Second, this study adds to the understanding of the social contagion theory in analyzing the role of intergroup bias in crypto owners' decisions in one of Southeast Asia's largest countries, Indonesia. Indonesia is identical to the collective community and young-age generation that is relevant to intergroup bias behavior and cryptocurrency investment. Third, this study provides a new understanding of the dual-system perspective by exploring two types of social environments: subjective norms and intergroup bias. Subjective norms and intergroup bias provided a strong, different impetus for cryptocurrency investment in adverse market conditions. Subjective norms have caused investors to refrain from investing in cryptocurrency over the past year. On the contrary, intergroup bias contributes to cryptocurrency investment even in declining market conditions. Fourth, the findings of overborrowing bias in cryptocurrency investment decisions open a new perspective in which crypto owners with overborrowing behavior have a tendency to act impulsively and irrationally, mainly when associated with a speculative investment in adverse market conditions. Finally, this study's practical implication is to provide government input to prevent vulnerable individual investors from buying or investing in cryptocurrencies.

5. Conclusions and Limitations

This study investigates whether intergroup bias, subjective norms, and self-control bias are predictors of crypto owners' investment decisions over the past year of the declining cryptocurrency market. Self-control bias in this study explores two types of behaviors: overborrowing and spending control. The results reveal that intergroup bias and overborrowing are the most impulsive factors contributing to the cryptocurrency investment decision over the past year, especially in the heaviest period. The empirical results indicate that intragroup bias due to the contagion effect from secondary groups of investors' social environments—for example, religious-based groups or sports clubs—encouraged investors to invest in the cryptocurrency market even though the market was in adverse conditions. Intergroup bias behavior that is more positive towards one's group members than those outside the group potentially results in irrational behavior, since the trust bias toward one's group influences the investment decision. The other finding is that overborrowing bias causes investors to behave irrationally, since instead of solving their debt problems, they prefer to spend their money on cryptocurrency investment in adverse market conditions.

In contrast, this study reveals that the subjective norm from the primary group of one's social environment—for example, peers, the most important persons—and the market price influence the decision not to invest in the adverse cryptocurrency market. The subjective norm factor indicates the reflective system, which is slow, controlled, and analytical in making investment decisions during significant cryptocurrency price declines. The different results between the influence of subjective norms, intergroup bias, and overborrowing biased behaviors explain that there is a dual-system perspective, reflexive and reflective, which investors experience simultaneously and which influences investment decisions. When the impulsive and reflexive system reacts most strongly, investors can generate irrational behavior and make irrational investment decisions. However, the reflective

perspective encourages rational behavior. Finally, spending control bias is unconfirmed as a predictor of cryptocurrency investment decisions.

This research has some limitations. First, the location of the crypto owner population cannot be determined. Alternatively, internet users are used as the population of crypto owners in this study. Since not all internet users are crypto owners, there is the possibility for differences between internet users and crypto owners. Second, with regard to the number of crypto owners that responded to this study, it is still necessary to gather additional samples from all over Indonesia in order for them to accurately represent cryptocurrency investors. Third, this study does not distinguish between investors who make direct or indirect investments through funding. Therefore, there is a potential for investment decisions to be biased due to the influence of fund managers. Finally, the model's ability to anticipate the decision not to invest in cryptocurrency is greater than its ability to predict the decision to invest. In addition, the results of this study must be interpreted with caution due to the possibility of other factors predicting the decision during a gloomy phase. Therefore, it is anticipated that future studies will enhance the predictive model by incorporating more variables that have the ability to affect the choice to invest in cryptocurrencies during a gloomy phase.

For future studies, our research recommends developing a model including other biased behaviors and investors' demographic variables that affect vulnerable decisions by cryptocurrency investors. Future research needs to explore the other dimension of bias behaviors, which are still extensive and should investigate the influence of biased behaviors on cryptocurrency investment decisions in international settings.

Author Contributions: Conceptualization, E.T. and J.T.; methodology, E.T.; software, S.E.H.; validation, E.T. and S.E.H.; formal analysis, E.T. and S.E.H.; investigation, S.E.H.; resources, R.W.; data curation, R.W.; writing—original draft preparation, E.T.; writing—review and editing, E.T.; visualization, R.W.; supervision, J.T.; project administration, R.W.; funding acquisition, J.T. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the Ministry of Education, Culture, Research, and Technology Republic of Indonesia, grant number SP DIPA-023.17.1.690523/2022 dated 17 November 2021.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data presented in this study are available on request from the corresponding author.

Acknowledgments: The authors would like to thank the Directorate General of Higher Education, Research and Technology Ministry of Education, Culture, Research, and Technology for providing the research grant. The authors also thank the Institute of Research and Community Service, Petra Christian University, for providing the research grant administration.

Conflicts of Interest: The authors declare no conflict of interest.

References

- AFM. 2022. Investing in Cryptos in The Netherlands. Available online: https://www.afm.nl/~{}/profmedia/files/onderwerpen/ cryptos/rapport-marktonderzoek-cryptocurrencies-eng.pdf (accessed on 14 September 2022).
- Ahmad, Maqsood. 2022. The role of cognitive heuristic-driven biases in investment management activities and market efficiency: A research synthesis. *International Journal of Emerging Markets. ahead-of-print*. [CrossRef]
- Ahmad, Maqsood, and Qiang Wu. 2022. Does herding behaviour matter in investment management and perceived market efficiency? Evidence from an emerging market. *Management Decision* 60: 2148–73. [CrossRef]
- Ahmed, Zeeshan, Shahid Rasool, Qasim Saleem, Mubashir Ali Khan, and Shamsa Kanwal. 2022. Mediating Role of Risk Perception Between Behavioural Biases and Investor's Investment Decisions. *SAGE Open* 12: 1–8. [CrossRef]
- Ajzen, Icek. 1991. The theory of planned behavior. Organizational Behavior and Human Decision Processes 50: 179–211. [CrossRef]
- APJII. 2022. Association of Indonesian Internet Service Providers: Indonesian Internet Profile Survey 2022. Available online: https://apjii.or.id/survei (accessed on 20 June 2022).
- Bakker, Laurens, Warren Hare, Hassan Khosravi, and Bojan Ramadanovic. 2010. A social network model of investment behaviour in the stock market. *Physica A: Statistical Mechanics and Its Applications* 389: 1223–29. [CrossRef]

- Baur, Dirk G., KiHoon Hong, and Adrian D. Lee. 2018. Bitcoin: Medium of exchange or speculative assets? *Journal of International Financial Markets, Institutions and Money* 43: 177–89. [CrossRef]
- Bhatt, Gita. 2022. Reimagining Money in the Age of Crypto And Central Bank Digital Currency. Available online: https://www.imf. org/en/Blogs/Articles/2022/09/01/reimagining-money-in-the-age-of-crypto-and-central-bank-digital-currency (accessed on 10 August 2022).
- Blau, Benjamin. 2017. Price dynamics and speculative trading in Bitcoin. *Research in International Business and Finance* 41: 493–99. [CrossRef]
- Blockchain Association of Indonesia. 2022. Indonesia Crypto Outlook Report 2021. Available online: https://asosiasiblockchain.co.id (accessed on 2 July 2022).
- Borgatti, Stephen, and Rob Cross. 2003. A relational view of information seeking and learning in social networks. *Management Science* 49: 432–45. [CrossRef]
- Brière, Marie, Kim Oosterlinck, and Ariane Szafarz. 2015. Virtual Currency, Tangible Return: Portfolio Diversification with Bitcoin. Journal of Law, Technology and the Internet 7: 79–94. [CrossRef]
- Chan, Kenneth, Jennifer Te Lai, and Tingting Li. 2022. Cultural values, genes and savings behaviour in China. *International Review of Economics and Finance* 80: 134–46. [CrossRef]
- Choi, Sujung. 2016. Herding among local individual investors: Evidence from online and offline trading. *Economics Letters* 144: 4–6. [CrossRef]
- CNBC Indonesia. 2022. Terungkap, Orang Indonesia Lebih Suka Kripto Dibanding Saham (Revealed, Indonesians Prefer Crypto Over Stocks). Available online: https://www.cnbcindonesia.com/market/20221110161922-17-386803/terungkap-orang-indonesia-lebih-suka-kripto-dibanding-saham (accessed on 25 December 2022).
- Cointelegraph. 2022. An Overview of the Cryptocurrency Regulations in Asia. Available online: https://cointelegraph.com/ cryptocurrency-regulation-for-beginners/an-overview-of-the-cryptocurrency-regulations-in-asia (accessed on 3 September 2022).
- Da Gama Silva, Paulo Vitor Jordão, Marcelo Cabus Klotzle, Antonio Carlos Figueiredo Pinto, and Leonardo Lima Gomes. 2019. Herding behaviour and contagion in the cryptocurrency market. *Journal of Behavioural and Experimental Finance* 22: 41–50. [CrossRef]
- De Dreu, Carsten, and Mariska Kret. 2016. Oxytocin Conditions Intergroup Relations Through Upregulated In-Group Empathy, Cooperation, Conformity, and Defense. *Biological Psychiatry* 79: 165–73. [CrossRef] [PubMed]
- Dobosh, Melissa. 2018. SAGE Research Methods: The SAGE Encyclopedia of Communication Research Methods. Thousand Oaks: SAGE Publications, Inc. [CrossRef]
- Eysenck, Sybil, and Hans Jürgen Eysenck. 1978. Impulsiveness and venturesomeness: Their position in a dimensional system of personality description. *Psychological Reports* 43: 1247–55. [CrossRef]
- Ferketich, Sandra. 1991. Focus on psychometrics: Aspects of item analysis. *Research in Nursing & Health* 14: 165–68. [CrossRef] Field, Andy. 2018. *Discovering Statistics Using IBM SPSS Statistics*, 5th ed. Los Angeles: Sage.
- Friehe, Tim, and Hannah Schildberg-Hörisch. 2017. Self-control and crime revisited: Disentangling the effect of self-control on risk taking and antisocial behaviour. *International Review of Law and Economics* 49: 23–32. [CrossRef]
- Friese, Malte, and Wilhelm Hofmann. 2009. Control me or I will control you: Impulses, trait self-control, and the guidance of behaviour. Journal of Research in Personality 43: 795–805. [CrossRef]
- Fujiki, Hiroshi. 2020. Who adopts crypto assets in Japan? Evidence from the 2019 financial literacy survey. *Journal of the Japanese and International Economies* 58: 101107. [CrossRef]
- Fujiki, Hiroshi. 2021. Crypto asset ownership, financial literacy, and investment experience. Applied Economics 53: 4560–81. [CrossRef]
- Fujino, Junya, Shisei Tei, Takashi Itahashi, Yuta Y. Aoki, Haruhisa Ohta, Manabu Kubota, Ryu-ichiro Hashimoto, Hidehiko Takahashi, Nobumasa Kato, and Motoaki Nakamura. 2020. Role of the right temporoparietal junction in intergroup bias in trust decisions. *Human Brain Mapping* 41: 1677–88. [CrossRef] [PubMed]
- Gailey, Alex, and Ryan Haar. 2022. The Future of Cryptocurrency: 8 Experts Share Predictions for the Second Half of 2022. Available online: https://time.com/nextadvisor/investing/cryptocurrency/future-of-cryptocurrency (accessed on 20 July 2022).
- Gathergood, John. 2012. Self-control, financial literacy and consumer over-indebtedness. *Journal of Economic Psychology* 33: 590–602. [CrossRef]
- Gemini. 2022. Global State of Crypto Report. Available online: https://www.gemini.com/gemini-2022-state-of-crypto-global.pdf (accessed on 10 July 2022).
- Google Finance. 2022. Bitcoin to Indonesian Rupiah. Available online: https://www.google.com/finance/quote/BTC-IDR?sa=X& ved=2ahUKEwjkp-yQhbr8AhXkzXMBHZ3ADj4Q-fUHegQIBhAe&window=1Y (accessed on 5 November 2022).
- Google, Temasek, Bain & Company. 2022. E-Conomy SEA 2022: Indonesia Report. Available online: https://economysea.withgoogle. com/intl/id_id/home/ (accessed on 23 December 2022).
- Gorodnichenko, Yuriy, and Gerard Roland. 2017. Culture, institutions and the wealth of nations. *Review of Economics and Statistics* 99: 402–16. [CrossRef]
- Harpe, Spencer. 2015. How to analyze Likert and other rating scale data. *Currents in Pharmacy Teaching and Learning* 7: 836–50. [CrossRef]
- Hosmer, David, Trina Hosmer, Saskia Le Cessie, and Stanley Lemeshow. 1997. A comparison of goodness-of-fit tests for the logistic regression model. *Statistics in Medicine* 16: 965–80. [CrossRef]

- Jakarta Globe. 2022. Indonesia Enjoys The Highest Crypto Adoption Rate Worldwide: Study. Available online: https://jakartaglobe. id/business/indonesia-enjoys-the-highest-crypto-adoption-rate-worldwide-study (accessed on 15 August 2022).
- Kasser, Tim, and Aaron Ahuvia. 2002. Materialistic values and well-being in business students. *European Journal of Social Psychology* 32: 137–46. [CrossRef]
- Kawamura, Tetsuya, Tomoharu Mori, Taizo Motonishi, and Kazuhito Ogawa. 2021. Is Financial Literacy Dangerous? Financial Literacy, Behavioural Factors, and Financial Choices of Households. *Journal of the Japanese and International Economies* 60: 101131. [CrossRef]
- Kinatta, Moses Munyami, Twaha Kigongo Kaawaase, John C.Kigozi Munene, Isaac Nabeta Nkote, and Stephen Korutaro Nkundabanyanga. 2022. Cognitive bias, intuitive attributes and investment decision quality in commercial real estate in Uganda. *Journal* of Property Investment and Finance 40: 165–87. [CrossRef]
- Kocher, Martin G, Konstantin E. Lucks, and David Schindler. 2019. Unleashing animal spirits: Self-control and overpricing in experimental asset markets. *The Review of Financial Studies* 32: 2149–78. [CrossRef]
- Kumar, Satish, Sandeep Rao, Kirti Goyal, and Nisha Goyal. 2022. Journal of Behavioural and Experimental Finance: A bibliometric overview. Journal of Behavioural and Experimental Finance 34: 100652. [CrossRef]
- Kumari, Sharda, Bibhas Chandra, and Jamini Kanta Pattanayak. 2020. Personality traits and motivation of individual users towards herding behaviour in Indian stock market. *Kybernetes* 49: 384–405. [CrossRef]
- Lei, Shan, and Leslie Ramos Salazar. 2022. Use of social networks in stock investment. *International Journal of Bank Marketing* 40: 110–27. [CrossRef]
- Liang, Quanxi, Jiangshan Liao, and Leng Ling. 2022. Social interactions and mutual fund portfolios: The role of alumni networks in China. *China Finance Review International* 12: 433–50. [CrossRef]
- Liu, Liu, and Hua Zhang. 2021. Financial literacy, self-efficacy and risky credit behaviour among college students: Evidence from online consumer credit. *Journal of Behavioral and Experimental Finance* 32: 100569. [CrossRef]
- McLeod, Saul. 2022. Correlation. Available online: https://www.simplypsychology.org/correlation.html (accessed on 24 August 2022).
- Neuner, Michael, Gerhard Raab, and Lucia Reisch. 2005. Compulsive buying in maturing consumer societies: An empirical re-inquiry. Journal of Economic Psychology 26: 509–22. [CrossRef]
- Osborne, Jason. 2017. Curvilinear effects in logistic regression. In *Best Practices in Logistic Regression*. London: SAGE Publications, Ltd., pp. 200–42. [CrossRef]
- Ouimet, Paige Parker, and Tate Geoffrey. 2020. Learning from Coworkers: Peer Effects on Individual Investment Decisions. *Journal of Finance* 75: 133–72. [CrossRef]
- Pallant, Julie. 2001. SPSS Survival Manual—A Step by Step Guide to Data Analysis Using SPSS for Windows. version 10. Buckingham: Open University Press.
- Peduzzi, Peter, John Concato, Elizabeth Kemper, Theodore Holford, and Alvan Feinstein. 1996. A simulation study of the number of events per variable in logistic regression analysis. *Journal of Clinical Epidemiology* 49: 1373–79. [CrossRef] [PubMed]
- Peng, Chao Ying Joanne, Kuk Lida Lee, and Gary Ingersoll. 2002. An Introduction to Logistic Regression Analysis and Reporting. *The Journal of Educational Research* 96: 3–14. [CrossRef]
- Peters, Linda D, Andrew D Pressey, and Wesley James Johnston. 2017. Contagion and learning in business networks. *Industrial Marketing Management* 61: 43–54. [CrossRef]
- Ryu, Hyun-Sun, and Kwang Sun Ko. 2019. Understanding speculative investment behaviour in the Bitcoin context from a dual-systems perspective. *Industrial Management and Data Systems* 119: 1431–56. [CrossRef]
- Santoso, Fendi, and Mohamad Ikhsan Modjo. 2022. Financial Literacy and Risk Tolerance of Indonesian Crypto-Asset Owners. Paper presented at the ICEME 2022: 13th International Conference on E-business, Management and Economics, Beijing, China, July 16–18; pp. 297–306. [CrossRef]
- Schober, Patrick, Christa Boer, and Lothar Schwarte. 2018. Correlation Coefficients: Appropriate Use and Interpretation. *Anesthesia & Analgesia* 126: 1763–68. [CrossRef]
- Sekita, Shizuka, Vikas Kakkar, and Masao Ogaki. 2022. Wealth, Financial Literacy and Behavioural Biases in Japan: The Effects of Various Types of Financial Literacy. *Journal of the Japanese and International Economies* 64: 101190. [CrossRef]
- Sekścińska, Katarzyna, Joanna Rudzinska-Wojciechowska, and Diana Jaworska. 2021. Self-control and investment choices. *Journal of Behavioural Decision Making* 34: 691–705. [CrossRef]
- Shah, Sayyed Sadaqat Hussain, Muhammad Asif Khan, Natanya Meyer, Daniel Francois Meyer, and Judit Oláh. 2019. Does herding bias drive the firm value? Evidence from the Chinese equity market. *Sustainability* 11: 5583. [CrossRef]
- Stix, Helmut. 2021. Ownership and purchase intention of crypto-assets: Survey results. Empirica 48: 65–99. [CrossRef]
- Suthar, Velo, Rohani Ahmad Tarmizi, Habshah Binti Midi, and Mohd Bakri Bin Adam. 2010. Students' beliefs on mathematics and achievement of university students: Logistics regression analysis. *Procedia-Social and Behavioural Sciences* 8: 525–31. [CrossRef]
- Tangney June Price, Roy Baumeister, and Angie Luzio Boone. 2004. High self-control predicts good adjustment, less pathology, better grades, and interpersonal success. *Journal of Personality* 72: 271–324. [CrossRef]
- Taylor, Shirley, and Peter Todd. 1995. Understanding information technology usage: A test of competing models. *Information Systems Research* 6: 144–76. [CrossRef]
- TripleA. 2022. Cryptocurrency Information about Indonesia. Available online: https://triple-a.io/crypto-ownership-indonesia/ (accessed on 24 December 2022).

- Westaby, James D, Danielle Pfaff, and Nicholas Redding. 2014. Psychology and social networks: A dynamic network theory perspective. *American Psychologist* 69: 269–84. [CrossRef]
- Yu, Haihong, Menghan Dan, Qingguo Ma, and Jia Jin. 2018. They all do it, will you? Event-related potential evidence of herding behavior in online peer-to-peer lending. *Neuroscience Letters* 681: 1–5. [CrossRef] [PubMed]
- Yue, Pengpeng, Aslihan Gizem Korkmaz, Zhichao Yin, and Haigang Zhou. 2022. The rise of digital finance: Financial inclusion or debt trap? *Finance Research Letters* 47: 102604. [CrossRef]
- Zhang, Shun, Yixuan Wang, Jieming Chen, and Cheng Cheng. 2018. Social network: An emerging income distribution mechanism in transitional urban China. *Journal of Administrative Sciences* 16: 83–108.
- Zhao, Haidong, and Lini Zhang. 2021. Financial literacy or investment experience: Which is more influential in cryptocurrency investment? *International Journal of Bank Marketing* 39: 1208–26. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





1

2

3

4

5

6 7

24

25

Article Rational Versus Irrational Behavior of Indonesian Cryptocurrency Owners in Making Investment Decisions

Elisa Tjondro *, Saarce Elsye Hatane, Retnaningtyas Widuri and Josua Tarigan

Department of Accounting, School of Business and Management, Petra Christian University, Siwalankerto 121-131, Surabaya 60236, Indonesia; <u>elsyehat@petra.ac.id</u> (SEH); <u>widuri@petra.ac.id</u> (RW); <u>josuat@petra.ac.id</u> (JT) * Correspondence: elisatjondro@petra.ac.id

Abstract: The purpose of this study is to investigate the salient factors that influence Indonesian 8 cryptocurrency owners in making their investment decision. This study employs intergroup bias, 9 subjective norms, overborrowing, and spending control to explain cryptocurrency investment be-10 havior. The questionnaire was collected from 309 respondents from the five largest internet user 11 areas, Jakarta, Surabaya, Bandung, Semarang, and Medan. This study executes the research frame-12 work using binary logistic regression. The results reveal that intergroup bias and overborrowing 13 are the most impulsive factors contributing to the cryptocurrency investment decision over the past 14 year. Furthermore, after November 2021, Indonesian crypto owners are more irrational in a bearish 15 period since their investment decisions are driven by their desire to be accepted in the social group. 16 Moreover, when they have overindebtedness, instead of solving their debt problems, they prefer to 17 spend their money on cryptocurrency investments. The subjective norms' influencers suggest that 18 crypto owners not invest when the cryptocurrency price sharp declining. The findings contribute to 19 the dual-systems perspective and social contagion theories enriching the empirical study regarding 20 investment behavior. 21

Keywords:cryptocurrency, intergroup bias, subjective norms, self-control, overborrowing, spend-22ing control, survey.23

1. Introduction

Cryptocurrency investors frequently act irrationally in making investment decisions. 26 This study explores how intergroup bias, subjective norms, and self-control factors influ-27 ence the rationality of investment decisions. Empirical evidence has shown that inves-28 tors do not always act rationally (Ahmad & Wu, 2022), including in the cryptocurrency 29 market. Previous research on the behavioral bias in the equity market (Kumari et al., 30 2020; Ahmad & Wu, 2022; Ahmad, 2022; Ahmed et al., 2022; Lei & Salazar, 2022; Liang et 31 al., 2022), commercial real estate (Kinatta et al., 2022), and cryptocurrency market (Ryu & 32 Ko, 2019). Ryu & Ko (2019) has exceptional research on cryptocurrency investment deci-33 sions, which show that strong impulse and weak self-control impact speculative bitcoin 34 investments. The study of behavioral bias can help understand individual investors 35 from different environments resulting in discrete investment decisions. An individual 36 has a common tendency to imitate, refer and observe other behavior, specifically in a 37 declining or unstable market condition (Yu et al., 2018; Shah et al., 2019). 38

In this study, there are four types of behavioral bias: intergroup bias, subjective norms,39overborrowing, and spending control. First, intergroup bias is a tendency to behave40more positively and provide greater rewards for their group members than outside41groups (De Dreu & Kret, 2016; Fujino et al., 2020). Intergroup bias in this study is fo-42cused on bias originating from a secondary group of investors' social environment that43is identical with lower intimacy, lower frequency, and duration of interaction, for44

Risks 2022, 10, x. https://doi.org/10.3390/xxxxx

example, religion-based groups or sports groups. Although group members have similar 45 interests, the members' purpose is to build social networks, bridging and bonding capi-46 tal (Lei & Salazar, 2022) that can increase the members' income and wealth status (Zhang 47 et al., 2018). Consistent with Chan et al. (2022) suggest that collectivist social values in-48 fluence individual financial behavior, due to a sense of solidarity in homogeneous com-49 munities. Furthermore, in a game task experiment, intergroup bias impacts individuals' 50 tendency to invest more in their group than in outside groups (Fujino et al., 2020). Indi-51 vidual behavior that is more positive towards their group members potentially results in 52 irrational investment decisions since the trust bias toward their group influences the in-53 vestment decision. The social contagion theory supports this argument (Bakker et al., 54 2010). Second, in the theory of planned behavior (TPB), subjective norm refers to beliefs 55 about the expectations or important references of peers and the most important persons 56 of the investors and motivates them to fulfill these expectations. Subjective norms in this 57 study focus on a primary group of investors' social environment: peers, the most im-58 portant persons, and the price trend. Subjective norm is a significant determinant influ-59 encing investment decisions, for instance, adopting and using technology (Ajzen, 1991). 60 Third, overborrowing reflects financial behavior related to high credit interest or exces-61 sive loans (Kawamura et al., 2021). Overborrowing is frequently associated with impul-62 sive behavior to buy or invest without thinking about the future. Investors associated 63 with high overborrowing behavior have a propensity for investing in cryptocurrency, 64 though in a high uncertainty period, support an irrational investment decision. Finally, 65 spending control bias is a compulsive buying behavior associated with unstable, self-66 inconsistent, negative emotions and perceptions of oneself (Liu & Zhang, 2021). Weak 67 spending control behavior can generate irrational investment decisions. However, stud-68 ies on behavior bias in a cryptocurrency investment decision, particularly intergroup 69 bias, subjective norms, and self-control bias still limited, so the reason motivates this 70 study. To fill this gap, this study aims to expand on Ryu and Ko (2019) by examining 71 whether intergroup bias, subjective norm, and self-control bias influence the investment 72 decisions whether to invest or not in the cryptocurrency market. 73

Over the past 2021 until the third quarter of 2022, cryptocurrency markets are facing 74 enormous challenges, with a very significant decline in market value, even though there 75 have been several small surges in the past few weeks. This study collected a survey from 76 crypto owners regarding their decisions during that period. The first half of 2022 was a 77 terrible period for the cryptocurrency market. Bitcoin and Ethereum, the two largest cryp-78 tocurrencies, declined more than 50 percent from their highs in November 2021 (Time, 79 2022). Based on figure 1, the Bitcoin market, compared to the Indonesian Rupiah, had de-80 creased by around 67% since its highest position on November 8, 2021, in the amount of 81 963 million to 314 million on August 20, 2022, when the data collection was performed. 82 The bearish market has the potential to influence investors' perceptions as market partic-83 ipants as well as the social environment with which investors interact. Then, it has an 84 impact on rational or irrational investors' behavior in cryptocurrency investment deci-85 sions. Uncertainty conditions generally lead to various positive or negative attitudes in 86 the social environment that can influence investors' investment decisions. Investors who 87 decided to invest or not invest in cryptocurrencies over the past year show that the dual-88 system perspective, which is a reflexive and reflective system runs in harmony in the de-89 cision-making process. The reflexive system is fast, impulsive, automatic, and uncon-90 scious, whereas the reflective system is slow, controlled, conscious, and analytical (Ryu & 91 Ko, 2019). Factors of intergroup bias, subjective norm, overborrowing, and spending con-92 93 trol bias can trigger the dual-system perspective, resulting in rational or irrational behaviors in the investment decision. 94

95



Figure 1. Bitcoin to Indonesian Rupiah from November 2021 to November 2022. Source: Google96Finance (2022).97

Positive or negative attitudes towards cryptocurrency investment originating from 98 the investors' social environment and create a more significant gap when the market is in 99 a declining condition. The cryptocurrency market has unique characteristics different 100 from conventional markets, for example, stocks and property markets. Cryptocurrencies 101 provide a new alternative investment. Individuals believe that digital money is the money 102 of the future (Bhatt, 2022), and the number of users is increasing progressively. However, 103 cryptocurrencies are risky speculative investments despite their inherent digital future 104 potential. Most Southeast Asia countries, including Indonesia, Malaysia, and Vietnam, 105 consider cryptocurrency illegal as a medium of exchange, though legal as an investment 106 or commodity. In addition, Thailand has just started to tighten the regulation of crypto-107 currencies (Cointelegraph, 2022). Therefore, it is critical for investors and potential inves-108 tors to understand the applicable regulations and make decisions with a complete under-109 standing of the potential risks. 110

Individual investors from the same geographic area were more likely to adopt biased 111 behavior than cross-country investors (Choi, 2016). Indonesia has seen a 280 percent 112 growth in the number of crypto investors since 2020, from 1.5 million to 4.2 million indi-113 viduals, with a daily trading volume reaching USD 117.4 million (Blockchain Association 114 of Indonesia, 2022). A study by Gemini (2022) entitled "Global State of Crypto Report" 115 found that 41 percent of Indonesians aged between 18 and 75 with an income of more than 116 \$14,000 per year own cryptocurrencies. The research also found that 61 percent of Indo-117 nesian respondents agree that crypto is the future of money which is the highest in the 118 Asia Pacific (Gemini, 2022). This study uses data from the Indonesian cryptocurrency mar-119 ket for three main reasons. First, there has been an acceleration of digital economic growth 120 in Indonesia after the COVID-19 pandemic. As the largest economy in Southeast Asia, 121 Indonesia has shown a significant increase in the value of the digital industry from US\$41 122 billion in 2019 to US\$ 77 billion in 2022. It is driven primarily by e-commerce (Google, 123 Temasek, Bain & Company, 2022). Digital financial services increase dominated by the 124 digital investment that increase to 31% CAGR (Compound Annual Growth Rate) in 2022 125 and an estimated 74% CAGR in 2025 (Google, Temasek, Bain & Company, 2022). These 126 data show the significant potential of Indonesia's digital investment in Southeast Asia. 127 Second, the number of individual Indonesia investors investing in cryptocurrency is 128 greater than those investing in stocks in 2022. As of June 2022, the number of cryptocur-129 rency investors was 15.1 million versus 9.1 million stock investors, despite the fact crypto 130 investment is still relatively new in Indonesia (CNBCIndonesia, 2022). Third, cryptocur-131 rency investors in Asia are dominated by the young generation (Fujiki, 2020; Fujiki, 2021; 132 Santoso & Modjo, 2022), so they fit the Indonesian demographic profile. Based on data for 133 2022, 78 percent of Indonesian crypto owners are aged 18 to 44 years (TripleA, 2022). 134 Therefore, the Indonesian market provides a unique setting for researchers to analyze the 135 influence of individual bias behavior on cryptocurrency investment decisions. 136

The findings of this study provide novel evidence supporting the dual-system per-137 spective and contagion theory by emphasizing the importance of understanding the in-138 fluence of intergroup bias, subjective norms, and self-control bias on investors' rational or 139 irrational behavior in the decision-making process. Recent studies investigate the effect of 140 subjective norms and self-control (Ryu & Ko, 2019), financial literacy and investment ex-141 perience (Zhao & Zhang, 2021; Fujiki, 2021), attitude and trust (Stix, 2021) in cryptocur-142 rency investment decisions. This research is different from the studies in the following 143 ways. First, although the Ryu and Ko (2019) study were conducted during the declining 144 market of cryptocurrencies, the Ryu and Ko (2019) study did not discuss intergroup bias 145 factor and did not analyze which factors determine investment decisions in cryptocurren-146 cies. Second, although Zhao and Zhang (2021), Fujiki (2021), and Stix (2021) found that 147 several factors were proven to influence crypto owners' investment decisions, their stud-148 ies did not address intergroup bias, subjective norms, and self-control bias as factors in-149 fluencing the decisions. Thus, this study is novel since this study demonstrates that inter-150 group bias and subjective norms result in different stimuli on crypto owners' investment 151 decisions. Intergroup bias contributes to the decision to invest in the market despite the 152 declining conditions. On the other hand, subjective norms contribute to the decision not 153 to invest in cryptocurrencies when market conditions experience a significant decline. 154 This study also finds that overborrowing can result in irrational behavior of investors to 155 keep investing in cryptocurrency in declining conditions. 156

This study contributes to the cryptocurrency literature in the following ways. 157 First, this study contributes to the development of cryptocurrency literature in Asia, which 158 is synonymous with a collectivist culture that is vulnerable to the contagion effect of in-159 vestment behavior. It provides empirical evidence supporting dual-system and social con-160 tagion theories by identifying intergroup bias, subjective norms, and overborrowing bias 161 as the impulsive factors contributing to the cryptocurrency investment decision. Second, 162 analysis of individual investors' biased behavior is performed in extreme declining period 163 which is still limited. During periods of significant market decline, the risk associated with 164 cryptocurrency investments for owners with vulnerable risks, such as contagion risk and 165 financial risk, increases. Therefore, irrational investor more inclined to invest in crypto-166 currency during adverse period. Finally, this study enhances the behavioral finance liter-167 ature on the rational and irrational behavior of crypto owners in making investment de-168 cisions by providing evidence of the effect of intergroup bias, subjective norms, and over-169 borrowing bias on cryptocurrency investment decisions. 170

The discussion of this study is then divided into several sections. Section 2 discusses 171 the literature review and hypothesis development. Section 3 describes the research methodology, including the sample selection and analysis model. Section 4 shows the results 173 of statistical tests, the interpretation of results, theoretical and practical implications. Finally, section 5 describes the conclusions, limitations of the study, and potential future 175 research. 176

2. Literature Review

177

177

2.1. Intergroup bias and Subjective Norms in Cryptocurrencies Investment Decision

Behavioral research of individual crypto owners, especially in emerging markets, is179an interesting topic and has broad future potential. Kumar et al. (2022), who conducted a180bibliometric study in the field of behavioral finance, suggest additional research is181needed to understand the factors that influence investors' behavior in the markets. According to Kumar et al. (2022), individual decision-makers differ fundamentally, contributing to differences in financial behavior in investment decision-making. The suggestion184from da Gama Silva et al. (2019) is to analyze bias behavior when the cryptocurrency185

market is in a sharp decline since it has the potential to provide new findings for the de-186 velopment of the literature. 187

This study discusses biased behavior with two focuses: intergroup bias and subjec-188 tive norm. First, intergroup bias is the tendency for members of one group to behave more 189 positively and provide greater rewards than those outside group (De Dreu & Kret, 2016; 190 Fujino et al. 2020). Collectivist culture has a strong positive influence on financial behavior 191 (Chan et al. 2022). The impact of intergroup bias on the investment market is more de-192 structive than the subjective norm because it involves more irrational, illogical thinking 193 and blindly imitates the actions of others because of psychological and emotional factors. 194 The need to be recognized as part of a social group can also lead to biased behavior in 195 investment decisions. Second, subjective norms are beliefs about the expectations or im-196 portant references of others that motivate investors to meet these expectations (Ajzen, 197 1991), for example, expectations from peers, the most important persons and references to 198 199 market trends. Crypto owners are inclined to follow advice from the closest social environment, including peers, the most important persons in making investment decisions, 200 and price trends. Crypto investors rely on peers to reduce their potential risk due to wrong 201 investment decisions. Bias investors try to match the investment performance of peers by 202 relying on others investment decisions. The two types of biased behaviors have different 203 motivations and produce different behaviors in investment decisions. 204

2.2. Hypothesis Development

2.2.1. Intergroup Bias and Contagion Effect in Collectivist Culture

Intergroup bias is a bias behavior of decision-making caused by the contagion effect 207 from a secondary social group. The contagion effect arises since investors have a high 208 level of trust (Bakker et al., 2010) towards other members of the social club through reg-209 ular interaction. The perspective of social contagion theory is relevant to investment de-210 cision since individuals tend to adopt similar behavior when they trust the information 211 provided by members of their social network (Westaby et al. 2014). The three character-212 istics of the social contagion effect that influence intergroup bias are being aware of the 213 knowledge that others have, appreciating what other person know, and gaining access 214 to one's thinking patterns (Borgatti, 2003). The convergence of attitudes and beliefs de-215 pends on exposure to information obtained when communicating between social net-216 works or groups (Peters et al., 2017). The interaction between members of a social group 217 or social network as bridging and bonding capital (Lei & Salazar, 2022) can increase the 218 members' wealth status. Furthermore, biased investors obtain financial knowledge, re-219 sources, and business opportunities from their groups, which in turn increase their in-220 come (Zhang et al., 2018). 221

Areas inhabited by various ethnic groups are one of the causes of massive contagion 222 effect behavior (Chan et al., 2022). Concern over one's immediate ethnic survival leads to 223 solidarity in genetically homogeneous communities, which is conducive to developing 224 collectivist cultures (Chan et al., 2022). Cultures based on certain groups or ethnicities, such as beliefs, norms, and social values, tend to remain or not easily change over a long 226 time (Gorodnichenko & Roland, 2017). The existence of various ethnic groups in Indonesia 227 is one of the motivating factors for research on intergroup bias. 228

H1: Intergroup bias contributes to the cryptocurrency investment decision over the past 230 year. 231

2.2.2. Subjective norm and Cryptocurrency Investment Decision

The subjective norm in this study focuses on the influence of the closest social environ-233 ment, which are peers and the most important persons, market trends on cryptocurrency 234

225

229

232

205

249

market. During an adverse period, influencers of subjective norms tend to suggest not to	235
invest in cryptocurrencies. They prefer to manage their risk exposure since cryptocur-	236
rency is a speculative investment (Ryu & Ko, 2019). Long-term goal, rational, and analyt-	237
ical are the driver of the influencers of subjective norms. The primary social group of	238
investors, for example: peers, the most important persons contribute to the decision to	239
avoid investing in cryptocurrency during a terrible period. Ouimet & Geoffrey (2020)	240
found that peers from the same employer firm influence an individual financial deci-	241
sions.	242

In contrast, subjective norms may also result in investors' impaired technical 243 knowledge and reasoning abilities, causing errors in judgment. Consequently, investors 244 make irrational decisions, which can adversely affect their returns (Ahmad & Wu, 2022). 245 Based on the arguments that have been explained, the research hypothesis is 246

H2: Subjective norms contributes to the cryptocurrency investment decision over the past year. 247

2.2.3. Self-control Behavior and Cryptocurrency Investment Decision

Self-control is the ability to regulate emotions and behavior and inhibit individual im-250pulses to achieve long-term results (Sekścińska et al., 2021). This study focuses on two251dimensions of self-control, overborrowing and spending control. Previous studies that252discussed financial behavior focused more on linking self-control with financial risk-253taking or gambling risks, but issues related to investment choice were mostly ignored254(Sekścińska et al., 2021). This study argues that overborrowing and spending controls255contribute to the cryptocurrency investment decision.256

Overborrowing reflects financial behavior synonymous with high credit interest or ex-257 cessive loans (Kawamura et al., 2021). Overborrowing is often associated with impulsive 258 behavior, whereas in decisions to buy or invest, individuals act without thinking about 259 the future. Gathergood (2012) showed that individuals who act impulsively tend to use 260 various types of credit, including consumer credit which makes them more vulnerable 261 to financial risk. In line with the studies by Friehe & Schildberg-Hörisch (2017) and 262 Kocher et al. (2019), who found that low self-control increases risk-taking in investment 263 decisions. Investors with a high level of debt tend to raise their risk exposure irrationally 264 and invest in high-risk and speculative investments based on their emotions. 265

The other type of self-control is spending control. Low spending control is when 266 individuals engage in impulsive consumer behavior or compulsive buying (Neuner et al., 267 2005). Liu & Zhang (2021) found that compulsive buying was associated with unstable, 268 self-inconsistent, negative emotions and perceptions of oneself. Furthermore, compulsive 269 buying is similar to individuals who focus on materialistic values as a strategy to alleviate 270 anxiety in response to insecurity symptoms (Kasser & Ahuvia, 2002). Cryptocurrency is a 271 speculative investment with high volatility and the potential for greater returns, thus 272 providing an impulse for individuals with low spending control. This study contends that 273 low spending control contributed to the investment of cryptocurrencies, in a highly 274 uncertainty period. In contrast, investors with high spending control will choose to refrain 275 from investing in cryptocurrencies in a declining period. 276

- H3a: Overborrowing behavior has contributed to the cryptocurrency investment decision 278 over the past year. 279
- H3b: Spending control has contributed to the cryptocurrency investment decision over the past year. 280

3. Methodology

282

3.1. Sample selection and questionnaire study

The sample of this study is 309 respondents who are active crypto owners in Indone-284 sia and are actively involved in social clubs. The period of distributing the questionnaire 285 link is for four weeks, from the beginning to the end of August 2022, to 997 respondents 286 who are members of a pooled database of a credible and trusted survey service organiza-287 tion. Individuals who fill out the survey will receive a GoPay/OVO voucher of IDR 10,000 288 or equivalent to USD 0.65. A total of 532 individuals did not have cryptocurrencies, so 289 they were excluded from the sample. Forty respondents were disqualified for filling out 290 the questionnaire incorrectly, twenty-five respondents did not complete filling out the 291 questionnaire, and ninety-one respondents were not actively involved in any social club. 292 This study uses several demographic criteria in sample selection to avoid sample selection 293 bias. First, the sample is an equal number of men and women. Second, the sample re-294 spondents came from five cities with a balanced number of 80 respondents per city with 295 a total of 400 respondents. Third, three age categories are the sample of this study, namely 296 21-30 years, 31-41 years, and 41-50 years. Respondents who did not fall into these three 297 categories were excluded from the sample. Finally, respondents were actively involved in 298 social clubs or clubs over the past year. A total of 91 respondents were not actively in-299 volved in social clubs or clubs; therefore, they were excluded from the sample. 300

This study conducted several stages of sample selection. First, this study accurately 301 and precisely specify the population. The population of crypto owners in Indonesia is dif-302 ficult to determine, and data are unavailable. Alternatively, this study uses the 2022 pop-303 ulation of internet users provided by the Association of Indonesian Internet Service Pro-304 viders (APJII). Previous studies have found that the characteristics of crypto owners are 305 active internet users at least once per week (Stix, 2021). Second, determining the sampling 306 frame phase. A sampling frame is all the available elements of a population that has a 307 chance of being selected for the survey (Dobosh, 2018). Respondents in this study are ver-308 ified members of a pooled database provided by a trusted surveyor service organization. 309 Individual targets are crypto owners who have a basic understanding and are actively 310 involved in social clubs. Respondent data based on the level of crypto knowledge is as 311 much as 67.6 percent have basic knowledge of cryptocurrency and 25.9 percent on the 312 level of thorough knowledge. Third, determine the sampling technique and use random 313 probability sampling. Probability sampling is when all elements of a sampling frame have 314 an equal chance of being selected for the sample. The use of random sampling can reduce 315 bias and increase the likelihood that the sample is representative (Dobosh, 2018). The 316 method of distributing survey was conducted online to 1,100 individuals who are verified 317 members of a database of surveyor services organizations so that all individuals have the 318 same opportunity to be selected as samples. 319

This study included a validation question in the form of simple mathematical addi-320 tion to ensure that respondents filled out the questionnaire consciously. If the respondent 321 answers incorrectly, it is assumed that the respondent was not fully conscious when an-322 swering the question, and the respondent is disqualified. There are five sections to the 323 questionnaire questions. First, demographic information and screening of current active 324 crypto owners. If the respondent does not have any cryptocurrencies, the respondent is 325 disqualified. Second, information regarding cryptocurrency acquirement over the past 326 year. The second part also asked about the subjective norm in cryptocurrency purchase 327 decisions. Third, the questions of intergroup bias begin with the definition of a social club 328 and whether the decision to participate actively in a social club. Respondents who an-329 swered that they had not been actively involved in a social club did not fill in this section 330 and were excluded from the sample. In the questionnaire, our study provides examples 331 of secondary social groups that are religious-based groups or sports clubs. This secondary 332 group's members are the same people for a set period, so interactions occur regularly ra-333 ther than just once or twice. Regular interaction with the social group is the main key to 334 building trust between members, which can cause a contagion effect on investment 335 behavior. Fourth, in this section, respondents were asked about the overborrowing experienced over the past year. Finally, the question of overborrowing and spending control. 337

The period of crypto ownership is the respondent who obtained crypto over the past 338 year. This study uses binary logistic regression and divides the dependent variable into 339 two groups of crypto owners. Respondents who obtained cryptocurrencies over the past 340 year are assigned number one, while those who do not meet the criteria are assigned num-341 ber two. A year cap was chosen to limit the current motivation that causes respondents to 342 buy cryptocurrencies. Stix (2021) stated that a more extended crypto buying period could 343 introduce research bias as motivations and influencing factors could potentially differ 344 from the study population. In addition, when the survey was conducted, the cryptocur-345 rency market conditions showed decreasing markets. Da Gama Silva et al (2019) also sug-346 gests analyzing biased behavior when the cryptocurrency markets are in a sharp decline 347 because it has the potential to provide new findings for the development of the literature. 348

This study uses binary logistic regression to analyze Indonesia's factors influencing 349 crypto ownership over the past year, especially in five big cities concentrated on Java and 350 Sumatra, Bandung, Semarang, Surabaya, Medan, and Jakarta. These five cities were chosen because the population of internet investors in these five cities represented 43.61% of 352 the population of internet investors in Indonesia, with an average internet penetration 353 ratio per province of 78.98% (APJII, 2022). Internet investors per province and provincial 354 capital are in table 1.

		Contribution	
The largest internet users	Capital of	to Indonesia's	Internet penetration
based on provinces	the province	total internet	ratio each province
		users	
West Java	Bandung	14.74%	82.4%
East Java	Surabaya	10.93%	72.9%
Central Java	Semarang	10.36%	76.9%
North Sumatra	Medan	4.34%	79.3%
DKI Jakarta	Jakarta	3.24%	83.4%
Total contribution nationally		43.61%	
Average internet			70 000/
penetration per province			10.90%

Table 1. The largest Internet users in Indonesia by province capital city.

¹Association of Indonesian Internet Service Providers (APJII), 2022

The sampling technique uses random sampling, and the sample size is determined 358 based on the number of variables, where ten observations are needed for each variable 359 studied. Peduzzi et al. (1996) and Peng (2002) use a minimum sample ratio of 10 to 1, with 360 a minimum sample size of 100. The formula is n = 10k/p, in which n is the number of 361 minimum samples, k is the number of predictors, and *p* is the smallest proportion of bi-362 nary cases in the population. The minimum sample size for a four-predictor model is 167; 363 thus, the sample size of 309 respondents considers meeting the requirement. Data collec-364 tion was carried out through surveys with online distribution in five provincial capitals 365 with the most significant internet investors in Indonesia. 366

There are several stages to preparing the instrument. First, the questionnaires from 367 previous references were translated into Indonesian and modified according to the research objectives. Second, the survey instrument was assessed by two experts: Professors 369 and practitioners in investment and accounting behavior. Third, a pilot project was held 370 first for 30 individuals not included in the research sample. Questionnaire questions that 371 do not pass the validity and reliability test will not be used in the survey. Finally, the 372 instrument was translated back into English for publication purposes. 373

356

392

405

3.2. Definition of variables and Model Analysis	374
Dependent variables	375
The dependent variable of this study is cryptocurrency investment over the past year	376
(PYI). The dependent variable is operationalized by a dummy variable, given a score of 1	377
if yes or a score of 2 otherwise. This study modifies the measurement of PYI from (Stix,	378
2021)'s study by focusing more on exploring the factors contributing to the decision to	379
investing cryptocurrency over the past year, especially during periods of extreme decline	380
in cryptocurrency markets.	381
Predictor variables	382
This study employs four predictor variables: intergroup bias, subjective norm, overbor-	383
rowing, and spending control bias. Predictor variables use five-point Likert scales. (Harpe,	384
2015) stated that the response measured by five-point Likert scales is continuous data, so	385
it is relevant to the independent variable in the logistic regression model. The intergroup	386
bias modified the questions from the study Kumari et al, (2020), and the subjective norm	387
predictor used modified the questionnaire (Taylor & Todd, 1995; Kumari et al, 2020; Kin-	388
atta et al, 2022). The instrument of overborrowing and spending control variables modi-	389
fying research questionnaire by (Eysenck & Eysenck, 1978; Tangney et al, 2004; Kawamura	390
et al, 2021; Sekita et al, 2022).	391

Demographic variables

Demographic variables in this study include gender, city of residence, age, occupation, 393 and activeness in social clubs. This study uses demographic variables as sample selection 394 criteria to avoid sample selection bias. First, the respondents were divided into two gen-395 der groups (Male and Female) with equal numbers (table 3). Second, the respondents 396 came from five cities, with a balanced number of eighty respondents for each area (table 397 3). Third, three age categories are the sample of this study, namely 21-30 years, 31-41 years, 398 and 41-50 years. Finally, respondents were actively involved in social clubs or clubs over 399 the past year. Several studies have found that demographic variables are associated with 400 the ownership of cryptocurrencies. For example, Fujiki (2020) found that crypto owners 401 in Japan are primarily male, under 30 years old, have a high pretax income, work in pri-402 vate or public companies, are the main source of income from a business and have a grad-403 uate school education level. 404

Table 2. Definition of variables and indicator

Construct	Indicators	Code
	Dependent variable	
Past year in- vestment	Invest in cryptocurrencies over the past year.	PYI
	Independent variables	
	The reason to invest in cryptocurrencies is to be recognised in a social group.	
	The reason to invest in cryptocurrencies is to follow the action of other group members.	
Intergroup bias	The reason to invest in cryptocurrencies is because of believing that other members have more knowledge about cryptocurren- cies.	IB
	The reason to invest in cryptocurrencies is the better perfor- mance of other group members.	

	Invest in a cryptocurrency whose value is rising in the market.			
Subjective norms	Investment decisions are based on the actions of others.			
	The reason to invest in cryptocurrencies is to follow the same pattern of decisions as other investors. The reason to invest in cryptocurrencies is that friends or coworkers believe that investing in cryptocurrencies is popu- lar.			
	The reason to invest in cryptocurrencies is that the most im- portant persons to me also invest in cryptocurrencies.			
	The reason to invest in cryptocurrencies is that people around me are doing so.			
	How frequently using consumer credit over the past year?			
	How frequently run out of money in your bank account over the past year?			
Overborrowing	How frequently have you had difficulty paying debts over the past year?	OB		
	How frequently have you borrowed money at extremely high- interest rates over the past year?			
Spending control	When making spending decisions, I carefully consider my fi- nancial situation.			
	When making a cryptocurrency investment decision, I try to spend my money wisely.			
	When making a cryptocurrency investment decision, I try to put in only a little time or effort.			

Research model:

406

415

416

 $\eta 1 = \eta \beta + \beta 1\xi 1 + \beta 2\xi 2 + \beta 3\xi 3 + \beta 4\xi 4 + \varepsilon$ (1)

Information	407
η1 = Past year investment (PYI)	408
$\eta\beta$ = Constant coefficient	409
$\beta 1\xi 1$ = Intergroup bias (IB)	410
$\beta 2\xi 2 = $ Subjective norms (SN)	411
β 3 ξ 3 = Overborrowing (OB)	412
$\beta 4\xi 4$ = Spending self-control (SPC)	413
$\mathcal{E} = \text{Error disturbance}$	414

4. Empirical Result

4.1. Demographic and Descriptive Statistics

In this study, 309 of the 400 crypto owners were actively involved in social clubs. 417 Table 3 presents the demographic of respondents, including the percentage of past year 418 investments (PYI), gender, age groups, type of occupation, and area. PYI describes re-419 spondents who invested in cryptocurrency over the past year as 35.90 percent of the total 420 respondents. The remaining 64.10 percent are crypto owners who did not invest in cryp-421 tocurrency. The age groups of respondents are between 21 to 50 years, whereas the age 422 category of 21 to 30 years dominates by 52.4%. Most respondents work in the private sec-423 tor (74.8 percent), followed by business owners at 17.2 percent in the next position. This 424 study conducts the Fisher exact test between gender and the dependent variable. The 425 result show an exact significant (2-sided) Fisher exact test value of 0.153 (> 0.05), confirm 426 that the sample is free from gender bias problems. 427

PastYear-Invest % Gender % Age % Occupation % % Area (PYI) Sample of 400 crypto owners Yes 30.0 47.8 20.0 Male 21-30 49.8 College student 4.3 Jabodetabek No 70.0 Female 52.2 31-40 37.3 Private employee 76.5 Surabaya 20.0 41-50 13.0 Business owner 15.3Semarang 20.0Full-time housewife 3.3 Bandung 20.0 Unemployment 0.8 Medan 20.0 A sample of 309 crypto owners actively involved in social clubs Yes 35.9 Male 4.5 23.3 44.021-30 52.4 College student Jabodetabek No 64.1 Female 56.0 31-40 38.2 Private employee 74.8 Surabaya 16.2 Semarang 41-50 9.4 Business owner 17.2 24.9 Full-time housewife 2.9 Bandung 23.9 Unemployment 0.6 Medan 11.7

Table 3. Demographic of respondents.

Table 4 shows the descriptive statistics of the dependent variable of PYI using cate-430gorical data, whereas the answer yes is given the number 1 and 2 otherwise. Independent431variables apply the mean score of the item indicators using a five-point Likert scale.432

Table 4. Descriptive Statistics.

	Ν	Mean	STD	Min	Max	VIF
PYI	309	1.641	0.481	1	2	
IB	309	3.435	0.823	1.00	5.00	1.882
SN	309	3.306	0.754	1.20	4.80	1.859
OB	309	1.965	0.795	1.00	5.00	1.015
SPC	309	4.123	0.656	1.00	5.00	1.035

4.2. Hypothesis Result

The hypothesis testing begins with the determination of the validity and reliability 435 of the indicators. The validity results using Pearson Correlation show coefficient values 436 between 0.614 to 0.889 (r> 0.60) for each item indicator so that it can be concluded that 437 item indicators can be used to measure the construct. The examination of reliability with 438 Cronbach's alpha shows that a value greater than 0.60 can be interpreted as high reliability 439 and an acceptable index (Pallant, 2001). Cronbach's alpha values were 0.807, 0.807, 0.728, 440 and 0.612 for IB, SN, OB, and SPC. The corrected item-total correlation ranged from 0.388 441 to 0.741, indicating good scales (Ferketich, 1991). The Pearson correlation results in table 442 5 display that the correlation coefficient between variables does not exceed 0.7. Thereby, 443 it can be concluded that there is no strong correlation between variables or is at a moderate 444 correlation level (Schober, 2018; McLeod, 2022). 445

Table 5. Pearson Correlation matrix.

	PYI	SN	IB	OB	SPC
PYI	1				
SN	0.191**	1			
IB	-0.028	0.675**	1		
OB	-0.207**	-0.008	0.025	1	
SPC	0.051	-0.004	0.104	-0.108	1

** *significant at the 0.01 and 0.05 level

428

433

429

434

A binary logistic regression was employed to determine the impact of intergroup 448 bias, subjective norms, and self-control derived from factor analysis on the crypto invest-449 ment decision. Table 6 shows the results of logit model with the Wald test. The empirical 450 result confirms that the intergroup bias (IB), subjective norms (SN), and overborrowing 451 (OB) factors were significant (Q<0.05) predictors of the odds of PYI. In contrast, the spend-452 ing control (SPC) factor was unconfirmed to predict the odds of PYI. Furthermore, the 453 influencers in the group of subjective norms (SN) do not suggest the crypto owners in-454 vesting during the heaviest period. On the contrary, intergroup bias (IB) and overborrow-455 ing (OB) have been predictors of cryptocurrency investment over the past year. This study 456 found that overborrowing bias (OB) has a stronger predictive ability to invest in crypto-457 currency than intergroup bias (IB). The exponential values of intergroup bias (IB) and 458 overborrowing (OB) are 0.423 and 0.576, respectively, indicating that the ability to predict 459 the odds ratio of PYI is greater in overborrowing (OB). The coefficient $\beta 1$ of intergroup 460 bias (IB) reveals that the odds ratio of investing in cryptocurrencies over the past year 461 decreases when the value of IB increases by one. The coefficient β 1 of the subjective norm 462 (SN) is 1.204, and the exponential coefficient is 3.333, meaning that the odds ratio of in-463 vestors who did not invest in cryptocurrencies over the past year increases by 3.333 times 464 as the value of SN increases by one when the other predictors are held constant. 465

Table 6. Coefficient of Predictor Factors.

	Dependent: PastYearInvestment (PYI)				
	β	S.E.	Wald	Sig.	Exp(β)
IB	-0.859	0.233	13.556	0.000	0.423
SN	1.204	0.251	23.095	0.000	3.333
OB	-0.551	0.162	11.577	0.001	0.576
SPC	0.227	0.198	1.311	0.252	1.254
Constant	-0.230	1.039	0.049	0.825	0.794

The goodness-of-fit statistics assess the fit of the logit model to the actual outcomes 467 (Peng et al, 2002). The omnibus test shows a significant model χ^2 (6) of 14.545 with a Q-468 value of 0.000 (Q<0.05) for the PYI model. The -2 log likelihood (-2LL) estimate measures 469 how well the estimated model fits with categorical data (Suthar et al, 2010). The value of 470 -2LL for the model is 363,157. Hosmer and Lemeshow's (Hosmer et al, 1997) test result 471 demonstrates a not significant value of 0.069 (Q>0.05). Thereby the model fit can be preserved. 473

Predicting a logit model accurately, including correctly predicting the outcome (Hosmer et al, 1997). Table 7 shows the ability to predict the PYI model of 73.463 percent. The ability to predict the crypto owners' decision not to invest in cryptocurrency (91.919 percent) is better than predicting investment decision over the past year (40.541 percent). The predictive ability of the model for cryptocurrency investment decisions (PYI = yes) is below 50 percent or weak. In other words, other factors not analyzed in the model influence investment decisions during the extreme declining period. 470

Table 7. Predicted Results.

			Predic	ted
	PastYearInvest (PYI)			
Observed		Yes	No	Percentage Correct
PastYearInves (PYI)	Yes	45	66	40.541
	No	16	182	91.919
Overall Percentage				73.463

Several assumptions must be met in logistic regression. First, the linearity 482 assumption uses box-tidwell transformation (Osborne, 2017; Field, 2018) to check for 483

466

489

linearity between predictors and the logit. The results of the linearity test reveal that the484box-tidwell transformation for the four independent variables is not significant to the485dependent variable, meaning that the linearity assumption is met. Second, the486multicollinearity test in table 4 shows the Variance Inflation Factor (VIF) value between4871.015 to 1.882 for the four independent variables.488

4.3. Discussion

The results from this study show that intergroup bias (IB) and overborrowing (OB) 490 behaviors are the most stimulating factors contributing to the cryptocurrency investment 491 decision in an adverse market. Both factors contribute to irrational behavior in making an 492 investment decision, specifically in the high uncertainty condition over the past year. In-493 vestors with the characteristics of having a high trust bias towards their social group and 494 high overborrowing behavior have a tendency to invest in cryptocurrency over the past 495 year when the price dropped by 67% from the highest point in November 2021 to the end 496 of August 2022 when data collection was performed. However, this study provides evi-497 dence that subjective norms (SN) of the primary social environment caused crypto owners 498 to refrain from investing in cryptocurrency over the past year. 499

Different types of the social environment have distinctive effects on cryptocurrency 500 investment decisions in a high-uncertainty market. The dual-system perspectives can ex-501 plain these distinctive effects, which discuss reflexive and reflective perspectives (Ryu & 502 Ko, 2019). Cryptocurrency investment, as a speculative investment activity, emerges as a 503 natural response to individual high and low-impulse interactions (Ryu & Ko, 2019). When 504 impulsive and reflexive investors react most strongly, investors can make irrational deci-505 sions. Nevertheless, a reflective perspective encourages rational behavior. These two im-506 pulses go hand in hand. Investors can receive different impulses of investing or not in-507 vesting in cryptocurrency simultaneously. Dual-system perspective, reflexive and reflec-508 tive, does not occur in isolation but side by side in speculative cryptocurrency invest-509 ments. Investors who decided to buy or not buy crypto over the past year show that the 510 reflexive and reflective system runs in harmony in the decision-making process, whether 511 influenced by intergroup bias or subjective norms. Consistent with da Gama Silva et al. 512 (2019) found that negative news in the cryptocurrency markets is related to behavior bias. 513

Intergroup bias from the secondary group of investors' social environment, such as 514 religion-based groups and sports groups, contributes to irrational behavior when making 515 an investment decision. There are two fundamental explanations related to intergroup 516 bias and irrational behavior. First, investors who are actively involved as members in a 517 social group tend to behave more positively, provide greater rewards, and have higher 518 trust in the group members than outside groups or known as trust bias. The trust bias 519 encourages investors to behave identically to their group members and make irrational 520 investment decisions since they want to be recognized in the group. Furthermore, inter-521 group bias encourages investors to act fast, impulsive, automatically, and unconsciously 522 in making investment decisions to obtain financial knowledge, resources, and business 523 opportunities from their social groups, which can increase their income. Interaction be-524 tween members of a social group or network as bridging and bonding capital (Lei & Sal-525 azar, 2022) can increase the members' income and wealth status (Zhang et al., 2018). Sec-526 ond, cryptocurrency investments provide different options because of their attractive 527 characteristics, high volatility, higher average returns, accessibility of weekend trading, 528 and low correlation with traditional assets. These characteristics are the advantages of 529 investment diversification (Brière et al., 2015). Then, the social group, which generally 530 prioritizes individual wealth status and exclusive networking, tends to stimulate the irra-531 tional behavior of crypto owners in making investment decisions. 532

Regarding to the subjective norms, this study shows that the investors received a 533 stimulus from their primary group of social environment, for example, peers, the most 534 important persons, and the price trends, not investing in cryptocurrency in the adverse 535

period. Cryptocurrency is a speculative investment instead of a long-term investment 536 (AFM, 2022). Indonesian regulations state that crypto is illegal as a medium of exchange. 537 However, it is allowed to be traded as a commodity (Jakarta Globe, 2022), thereby expand-538 ing its function as a speculative investment. Previous studies also confirmed that Bitcoin 539 is mainly used as a speculative asset rather than an alternative currency (Blau, 2017; Baur 540 et al., 2018). Thus, in a high uncertainty condition, persons in the closest social environ-541 ment of crypto owners tend to act cautious, slow, controlled, conscious, and analytically, 542 exposing the reflective system. They try to convince crypto owners not to invest in cryp-543 tocurrency during the heaviest period. 544

Besides intergroup bias, the other impetus to invest in cryptocurrency in the declin-545 ing market arises from overborrowing bias. The reason is that individuals with low self-546 control often act on a reflexive perspective, leading to high levels of unplanned (Friese & 547 Hofmann, 2019) and irrational behavior. Consistent with Ryu & Ko (2019) stated that 548 strong impulses and weak self-control drive speculative investment behavior in the cryp-549 tocurrency context. Easy access to fintech credit markets increases the risk of individuals 550 falling into debt traps (Yue et al., 2022). Liu & Zhang (2021) explained that easy access to 551 online consumer credit became one of the causes of severe financial risk. The digital credit 552 market trap is a challenge faced by crypto owners, who generally always come into con-553 tact with digital media. 554

The theoretical and practical implications of this study are described in several sec-555 tions. First, survey studies on biased behavior and cryptocurrency investment decisions 556 are minimal, so this study enriches the literature on the irrational decisions of crypto own-557 ers, especially in Asia. Several crypto owners' studies that are relevant to this study in-558 clude Fujiki (2021) in Asia, Stix (2021) in Europe, and Zhao & Zhang (2021) in the USA. 559 More specifically, studies with a sample of crypto owners in the Asia region have yet to 560 receive much attention. Second, this study adds to the understanding of the social conta-561 gion theory in analysing the role of intergroup bias in crypto owners' decisions in one of 562 Southeast Asia's largest countries, Indonesia. Indonesia is identical to the collective com-563 munity and young age generation that is relevant to intergroup bias behavior and cryp-564tocurrency investment. Third, this study provides a new understanding of the dual-sys-565 tem perspective by exploring two types of social environments: subjective norms and in-566 tergroup bias. Subjective norms and intergroup bias provided a strong different impetus 567 for cryptocurrency investment in adverse market conditions. Subjective norms have 568 caused investors to refrain from investing in cryptocurrency over the past year. Contrary, 569 intergroup bias contributes to cryptocurrency investment even in declining market con-570 ditions. Fourth, the findings of overborrowing bias in cryptocurrency investment deci-571 sions open a new perspective that crypto owners with overborrowing behavior have a 572 tendency to act impulsively and irrationally, mainly when associated with a speculative 573 investment in adverse market conditions. Finally, this study's practical implication is to 574 provide government input to prevent vulnerable individual investors from buying or in-575 vesting in cryptocurrencies. 576

5. Conclusions and Limitations

This study investigates whether intergroup bias, subjective norm, and self-control 578 bias are predictors of crypto owners' investment decisions over the past year of the de-579 clining cryptocurrency market. Self-control bias in this study explores two types behav-580 iors, overborrowing and spending control. The results reveal that intergroup bias and 581 overborrowing are the most impulsive factors contributing to the cryptocurrency invest-582 ment decision over the past year, especially in the heaviest period. The empirical results 583 indicate that intragroup bias due to the contagion effect from secondary group of inves-584 tors social environment, for example religious-based groups or sports clubs, encouraged 585 investors to invest in the cryptocurrency market even though the market is in adverse 586 conditions. Intergroup bias behavior that is more positive towards their group members 587

than outside group potentially results in irrational behavior since the trust bias toward 588 their group influences the investment decision. The other finding is overborrowing bias 589 causes investors to behave irrationally since instead of solving their debt problems, they 590 prefer to spend their money on cryptocurrency investment in adverse market conditions. 591

In contrast, this study reveals that subjective norm from primary group of social en-592 vironment, for example peers, the most important persons, and market price influences 593 the decision not to invest in the adverse cryptocurrency market. The subjective norm fac-594 tor indicates the reflective system that is slow, controlled, and analytical in making invest-595 ment decisions during significant cryptocurrency price declines. The different result be-596 tween influence of subjective norm, intergroup bias and overborrowing biased behaviors 597 explain that there is a dual-system perspective, reflexive and reflective, which investors 598 experience simultaneously and influence investment decisions. When impulsive and re-599 flexive system reacts most strongly, investors can generate irrational behavior and make 600 irrational investment decisions. However, the reflective perspective encourages rational 601 behavior. Finally, spending control bias is unconfirmed as a predictor of cryptocurrency 602 investment decisions. 603

This research has some limitations. First, the location of the crypto owner population 604 cannot be determined. Alternatively, internet users are used as the population of crypto 605 owners in this study. Since not all internet users are crypto owners, there is the possibility 606 for differences between internet users and crypto owners. Second, with regard to the num-607 ber of crypto owners that responded to this study, it is still necessary to gather additional 608 samples from all over Indonesia in order for them to accurately represent cryptocurrency 609 investors. Third, this study does not distinguish between investors who make direct or 610 indirect investments through funding. Therefore, there is a potential for investment deci-611 sions to be biased due to the influence of fund managers. Finally, the model's ability to 612 anticipate the decision not to invest in cryptocurrency is greater than its ability to predict 613 the decision to invest. In addition, the results of this study must be interpreted with cau-614 tion due to the possibility of other factors in predicting the decision during a gloomy 615 phase. Therefore, it is anticipated that future studies will enhance the predictive model by 616 incorporating more variables that have the ability to affect the choice to invest in crypto-617 currencies during a gloomy phase. 618

For future study, our research recommends developing a model including other biased behavior and investors' demographic variables that affect vulnerable decisions by cryptocurrency investors. Future research needs to explore the other dimension of bias behaviors which are still extensive and should investigate the influence of biased behaviors on cryptocurrency investment decisions in international settings. 621 622 623

Author Contributions: E.T Prepare original manuscript, research background, literature review,
conceptual model, methodology; S.E.H using SPSS software, supervising processing, validation, re-
liability analysis, and hypothesis analysis; R.W descriptive statistics testing, final article analysis
and analysis. J.T Supervision, correcting writing errors. Authorship must be limited to those who
have con-tributed substantially to the work reported.624
625

Funding: This research was funded by DIPA Directorate General of Higher Education, Research629and Technology Ministry of Education, Culture, Research, and Technology, grant number SP DIPA-630023.17.1.690523/2022.631

Data Availability Statement: The data presented in this study are available on request from the632corresponding author.633

Acknowledgments: The authors would like to thank the Directorate General of Higher Education,634Research and Technology Ministry of Education, Culture, Research, and Technology for providing635the research grant. The authors also thank the Institute of Research and Community Service Petra636Christian University for providing research grant administration.637

Conflicts of Interest: The authors declare no conflict of interest.

References	640
Ahmad, M. The role of cognitive heuristic-driven biases in investment management activities and market efficiency: a research syn-	641
thesis. International Journal of Emerging Markets 2022, forthcoming. https://doi.org/10.1108/IJOEM-07-2020-0749	642
Ahmad, M.; Wu, Q. Does herding behaviour matter in investment management and perceived market efficiency? Evidence from an	643
emerging market. Management Decision 2022, 60, 2148–2173. <u>https://doi.org/10.1108/MD-07-2020-0867</u>	644
Ahmed, Z.; Rasool, S.; Saleem, Q.; Khan, M. A.; Kanwal, S. Mediating Role of Risk Perception Between Behavioural Biases and In-	645
vestor's Investment Decisions. SAGE Open 2022, 12. https://doi.org/10.1177/21582440221097394	646
APJII. Association of Indonesian Internet Service Providers: Indonesian Internet Profile Survey 2022. Available online:	647
https://apjii.or.id/survei (accessed on 20 June, 2022).	648
Bakker, L.; Hare, W.L.; Khosravi, H.; Ramadanovic, B. A social network model of investment behaviour in the stock market", Physica	649
A: Statistical Mechanics and its Applications 2010, 389, 1223-1229, doi: 10.1016/j.physa.2009.11.013	650
Baur, D.G.; Hong, K.; Lee, A.D. Bitcoin: medium of exchange or speculative assets? Journal of International Financial Markets, Insti-	651
tutions and Money 2018, 43, 177-189. https://doi.org/10.1016/j.intfin.2017.12.004	652
Bhatt, G. Reimagining Money In The Age Of Crypto And Central Bank Digital Currency. Available online:	653
https://www.imf.org/en/Blogs/Articles/2022/09/01/reimagining-money-in-the-age-of-crypto-and-central-bank-digital-currency	654
(accessed on 10 August 2022).	655
Blau, B.M. Price dynamics and speculative trading in Bitcoin. Research in International Business and Finance 2017, 41, 493-499.	656
https://doi.org/10.1016/j.ribaf.2017.05.010	657
Blockchain Association of Indonesia. Indonesia Crypto Outlook Report 2021. Available online: https://asosiasiblockchain.co.id (ac-	658
cessed on 2 July 2022).	659
Borgatti, S.P.; Cross, R. A relational view of information seeking and learning in social networks. Management Science 2003, 49, 432-	660
445. https://doi.org/10.1287/mnsc.49.4.432.14428	661
Brière, M.; Oosterlinck, K.; Szafarz, A. Virtual Currency, Tangible Return: Portfolio Diversification with Bitcoin. Journal of Law,	662
Technology and the Internet 2015 , 7, 79–94. <u>https://doi.org/10.1057/jam.2015.5</u>	663
Chan, K. S.; Lai, J. te.; Li, T. Cultural values, genes and savings behaviour in China. International Review of Economics and Finance	664
2022, 80, 134–146. https://doi.org/10.1016/j.iref.2022.02.009.	665
Choi, S. Herding among local individual investors: Evidence from online and offline trading. Economics Letters 2016, 144, 4-6.	666
https://doi.org/10.1016/j.econlet.2016.04.030.	667
CNBC Indonesia. Terungkap, Orang Indonesia Lebih Suka Kripto Dibanding Saham (Revealed, Indonesians Prefer Crypto Over	668
Stocks). Available online: https://www.cnbcindonesia.com/market/20221110161922-17-386803/terungkap-orang-indonesia-	669
lebih-suka-kripto-dibanding-saham (accessed on 25 December 2022).	670
Cointelegraph. An overview of the cryptocurrency regulations in Asia. Available online: https://cointelegraph.com/cryptocurrency-	671
regulation-for-beginners/an-overview-of-the-cryptocurrency-regulations-in-asia (accessed on 3 September 2022).	672
Da Gama Silva, P. V. J.; Klotzle, M. C.; Pinto, A. C. F.; Gomes, L. L. Herding behaviour and contagion in the cryptocurrency market.	673
Journal of Behavioural and Experimental Finance 2019, 22, 41–50. https://doi.org/10.1016/j.jbef.2019.01.006	674
Eysenck, S.B.; Eysenck H.J. Impulsiveness and venturesomeness: their position in a dimensional system of personality description.	675
Psychological reports 1978, 43, 1247 – 1255. https://doi.org/10.2466/pr0.1978.43.3f.1247	676
Ferketich, S. Focus on psychometrics: Aspects of item analysis. Research in Nursing & Health 1991, 14, 165–168.	677
<u>https://doi.org/10.1002/nur.4770140211</u>	678
Field, A. Discovering statistics using IBM SPSS statistics (5th ed). Los Angeles: Sage, 2018.	679
AFM. Investing in Cryptos in the Netherlands. Available online: https://www.afm.nl/~/profmedia/files/onderwerpen/cryptos/rap-	680
port-marktonderzoek-cryptocurrencies-eng.pdf (accessed on 14 September 2022).	681
Friene, I.; Schildberg-Horisch, H. Self-control and crime revisited: Disentangling the effect of self-control on risk taking and antisocial	682
behaviour. International Review of Law and Economics 2017, 49, 23–32. https://doi.org/10.1016/j.irie.2016.11.001.	683
Friese, M.; Hotmann, W. Control me or I will control you: impulses, trait self-control, and the guidance of behaviour. Journal of	684
Research in Personality 2009, 43, 795-805. https://doi.org/10.1016/j.jrp.2009.07.004	685
Fujiki, H. Crypto asset ownership, infancial interacy, and investment experience. Applied Economics 2021, 55, 4560–4581.	686
nttps://doi.org/10.1080/00036846.2021.1904125.	687
tional Economics 2020 58, 101107, https://doi.org/10.1016/j.iiis.2020.101107	600
Cothergood I. Solf control financial literacy and concurrence over indebtedness. Journal of Economic Psychology 2012, 22, 500, 602	609
bttps://doi.org/10.1016/j.joop.2011.11.006	601
Comini Clobal State of Crunto Report Available online: https://www.gomini.com/gomini.2022.state of grunte global ndf (accessed	602
on 10 July 2022)	6072
Coorde Temasek Bain & Company E-Conomy SEA 2022: Indonesia Report Available online: https://oconomy.coa.with	601
google com/intl/id_id/home/ (accessed on 23 December 2022)	695
Gorodnichenko Y · Roland G. Culture institutions and the wealth of nations. Review of Economics and Statistics 2017. 99. 402-416	696
https://doi.org/10.1162/REST a 00599	697

- Harpe, S. E. How to analyze Likert and other rating scale data. In Currents in Pharmacy Teaching and Learning 2015, 7, 836–850. 698 https://doi.org/10.1016/j.cptl.2015.08. 699
- Hosmer, D.W.; Hosmer, T.; Le Cessie, S.; Lemeshow, S. A comparison of goodness-of-fit tests for the logistic regression model. Sta-700 tistics in Medicine 1997, 16, 965-980, doi: 10.1002/(SICI)1097-0258(19970515)16:9<965::AID-SIM509>3.0.CO;2-O 701
- Jakarta Globe. Indonesia Enjoys The Highest Crypto Adoption Rate Worldwide: Study. Available online: https://jakartaglobe.id/busi-702 ness/indonesia-enjoys-the-highest-crypto-adoption-rate-worldwide-study (accessed on 15 August 2022). 703
- Kasser, T.; A. Ahuvia. Materialistic values and well-being in business students. European Journal of Social Psychology 2002, 32, 137-146. https://doi.org/10.1002/ejsp.85
- Kawamura, T.; Mori, T.; Motonishi, T.; Ogawa, K. Is Financial Literacy Dangerous? Financial Literacy, Behavioural Factors, and 706 Financial Choices of Households. Journal of the Japanese and International Economies 2021, 60, 101131. 707 https://doi.org/10.1016/j.jjie.2021.101131 708
- Kinatta, M. M.; Kaawaase, T. K.; Munene, J. C.; Nkote, I.; Nkundabanyanga, S. K. Cognitive bias, intuitive attributes and investment 709 decision quality in commercial real estate in Uganda. Journal of Property Investment and Finance 2022, 40, 165-187. 710 https://doi.org/10.1108/JPIF-11-2020-0129 711
- Kocher, M. G.; Lucks, K. E.; Schindler, D. Unleashing animal spirits: Self-control and overpricing in experimental asset markets. The 712 Review of Financial Studies 2019, 32, 2149–2178. https://doi.org/10.1093/rfs/hhy109. 713
- Kumar, S.; Rao, S.; Goyal, K.; Goyal, N. Journal of Behavioural and Experimental Finance: A bibliometric overview. In Journal of 714 Behavioural and Experimental Finance 2022, 34, 100652. https://doi.org/10.1016/j.jbef.2022.100652 715
- Kumari, S.; Chandra, B.; Pattanayak, J. K. Personality traits and motivation of individual users towards herding behaviour in Indian 716 stock market. Kybernetes 2020, 49, 384-405. https://doi.org/10.1108/K-11-2018-0635 717
- Lei, S.; Salazar, R.L. Use of social networks in stock investment. International Journal of Bank Marketing 2022, 40, 110-127. 718 https://doi.org/10.1108/IJBM-04-2021-0158 719
- Liang, Q.; Liao, J.; Ling, L. Social interactions and mutual fund portfolios: the role of alumni networks in China. China Finance Review International 2022, 12, 433–450. https://doi.org/10.1108/CFRI-04-2021-0073.
- Liu, L.; Zhang, H. Financial literacy, self-efficacy and risky credit behaviour among college students: Evidence from online consumer credit. J. Behav. Exp. Finance 2021, 32, 100569. http://dx.doi.org/10.1016/j.jbef.2021.100569.

McLeod, S. A. Correlation. Available online: https://www.simplypsychology.org/correlation.html (accessed on 24 August 2022).

- Neuner, M.; Raab, G.; Reisch, L. A. Compulsive buying in maturing consumer societies: An empirical re-inquiry. Journal of Economic 725 Psychology 2005, 26, 509-522. https://doi.org/10.1016/j.joep.2004.08.002 726
- Osborne, J. W. Curvilinear Effects in Logistic Regression, in Best Practices in Logistic Regression. SAGE Publications, Ltd, 2017; pp. 200-242. https://doi.org/10.4135/9781483399041.n7
- Ouimet, P.; Geoffrey, T. Learning from Coworkers: Peer Effects on Individual Investment Decisions. Journal of Finance 2020, 75, 133-172. https://doi.org/10.1111/jofi.12830
- Pallant, J. SPSS survival manual a step by step guide to data analysis using SPSS for windows (version 10). Buckingham Open University Press, 2001.
- Peduzzi, P; Concato, J; Kemper, E; Holford, T.R; Feinstein, A.R. A simulation study of the number of events per variable in logistic regression analysis 1996, 49, 1373-1379. https://doi.org/10.1016/S0895-4356(96)00236-3
- Peng, C.J.; Lee, K.L.; Ingersoll, G. M. An Introduction to Logistic Regression Analysis and Reporting. The Journal of Educational Research 2002, 96, 3-14. https://doi.org/10.1080/00220670209598786.
- Peters, L.D.; Pressey, A.D.; Johnston, W.J. Contagion and learning in business networks. Industrial Marketing Management 2017, 61, 43-54. https://doi.org/10.1016/j.indmarman.2016.06.011
- Ryu, H. S.; Ko, K. S. Understanding speculative investment behaviour in the Bitcoin context from a dual-systems perspective. Industrial Management and Data Systems 2019, 119, 1431-1456. https://doi.org/10.1108/IMDS-10-2018-0441
- Santoso, F.; & Modjo, M. I. Financial Literacy and Risk Tolerance of Indonesian Crypto-Asset Owners. ICEME 2022: 13th International Conference on E-business, Management and Economics 2022, 297 – 306. https://doi.org/10.1145/3556089.3556119
- Schober, P. Correlation Coefficients: Appropriate Use and Interpretation. Anesthesia & Analgesia 2018, 126, 1763-1768. doi: 10.1213/ANE.00000000002864.
- Sekścińska, K.; Rudzinska-Wojciechowska, J.; Jaworska, D. Self-control and investment choices. Journal of Behavioural Decision 745 Making 2021, 34, 691-705. https://doi.org/10.1002/bdm.2236.
- Sekita, S.; Kakkar, V.; Ogaki, M. Wealth, Financial Literacy and Behavioural Biases in Japan: the Effects of Various Types of Financial 747 Literacy. Journal of the Japanese and International Economies 2022, 64, 101190. https://doi.org/10.1016/j.jjie.2021.101190 748
- Shah, S. S. H., Khan, M. A., Meyer, N., Meyer, D. F., & Oláh, J. (2019). Does herding bias drive the firm value? Evidence from the 749 Chinese equity market. Sustainability, 11(20), 5583. https://doi.org/10.3390/su11205583
- Stix, H. Ownership and purchase intention of crypto-assets: survey results. Empirica 2021, 48, 65–99. https://doi.org/10.1007/s10663-751 020-09499-x. 752
- Suthar, V.; Tarmizi, R.A.; Midi, H.; Adam, M.B. Students' beliefs on mathematics and achievement of university students: logistics 753 regression analysis. Procedia-Social and Behavioural Sciences 2010, 8, 525-531. https://doi.org/10.1016/j.sbspro.2010.12.072 754
- Tangney, J.P.; Boone, A.L.; Baumeister, R.F. High self-control predicts good adjustment, less pathology, better grades, and interper-755 sonal success. Journal of Personality 2004, 72, 271-324. DOI: 10.1111/j.0022-3506.2004.00263.x 756

704

705

720

721

722

723

724

727

728

729 730

731

732

733

734

735

736

737

738

739

740

741

742

743

744

746

Taylor, S.; Todd, P.A. Understanding information technology usage: a test of competing models. Information Systems Research 1995,	757
6, 144-176. https://doi.org/10.1287/isre.6.2.144	758
TripleA. Cryptocurrency information about Indonesia. Available online: https://triple-a.io/crypto-ownership-indonesia/ (accessed on	759
24 December 2022).	760
Westaby, J.D.; Pfaff, D.L.; Redding, N. Psychology and social networks: a dynamic network theory perspective. American Psycholo-	761
gist 2014, 69, 269-284. https://doi.org/10.1037/a0036106	762
Yue, P.; Korkmaz, A. G.; Yin, Z.; Zhou, H. The rise of digital finance: Financial inclusion or debt trap? Finance Research Letters 2022,	763
47, 102604. https://doi.org/10.1016/j.frl.2021.102604.	764
Yu, H., Dan, M., Ma, Q., & Jin, J. (2018). They all do it, will you? Event-related potential evidence of herding behavior in onlinepeer-	765
to-peer lending. Neuroscience Letters, 681(1), 1–5. <u>https://doi.org/10.1016/j.neulet.2018.05.021</u>	766
Zhao, H.; Zhang, L. Financial literacy or investment experience: which is more influential in cryptocurrency investment? Interna-	767
tional Journal of Bank Marketing 2021, 39, 1208–1226. https://doi.org/10.1108/IJBM-11-2020-0552.	768
Zhang, S.; Wang, Y.; Chen, J.; Cheng, C. Social network: an emerging income distribution mechanism in transitional urban China.	769
Journal of Administrative Sciences 2018, 16, 83-108. https://dergipark.org.tr/tr/download/article-file/698560	770
	771